

# Temi di Discussione

(Working Papers)

Evaluating the efficacy of European regional funds for R&D

by Davide Fantino and Giusy Cannone







# Temi di discussione

(Working papers)

Evaluating the efficacy of European regional funds for R&D

by Davide Fantino and Giusy Cannone

Number 902 - February 2013

The purpose of the Temi di discussione series is to promote the circulation of working papers prepared within the Bank of Italy or presented in Bank seminars by outside economists with the aim of stimulating comments and suggestions.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: Massimo Sbracia, Stefano Neri, Luisa Carpinelli, Emanuela Ciapanna, Francesco D'Amuri, Alessandro Notarpietro, Pietro Rizza, Concetta Rondinelli, Tiziano Ropele, Andrea Silvestrini, Giordano Zevi. Editorial Assistants: Roberto Marano, Nicoletta Olivanti.

ISSN 1594-7939 (print) ISSN 2281-3950 (online)

Printed by the Printing and Publishing Division of the Bank of Italy

## EVALUATING THE EFFICACY OF EUROPEAN REGIONAL FUNDS FOR R&D

## by Davide Fantino<sup>\*</sup> and Giusy Cannone<sup>\*</sup>

## Abstract

This paper provides some empirical evidence of the impact of two policy measures designed to support innovation in small and medium firms in an Italian region, both financed using the European Structural Funds but managed at regional level. The first measure was a concessional loan to promote the introduction of innovative plant, machinery and equipment, while the second was a free grant to stimulate research activity by firms. The programmes were effective in stimulating targeted investments (respectively tangible and intangible), but the benefits were short-lived, although to different degrees. The impact was stronger for the smallest firms and, in the case of the second measure, for firms with a low credit rating.

**JEL Classification**: O32, O38. **Keywords**: R&D, public policy, evaluation.

### Contents

1. Introduction	
2. Description of the policy	7
3. Empirical strategy and data	
3.1 Empirical strategy	
3.2 Data	
4. Empirical evidence	
4.1 Investment	
4.2 Robustness checks	
4.3 Other outcome variables	
5. Effect on some subsamples	
6. Conclusions	
Tables	
References	

<sup>&</sup>amp; Bank of Italy, Economic Outlook and Monetary Policy Department.

<sup>▲</sup> Polytechnic University of Turin.

### **1. Introduction**<sup>\*</sup>

Public intervention is usually crucial to foster innovation and economic development. There exists a gap between the optimal social level of R&D investment and the private level, causing a systematic underprovision that could hamper a country's economic growth. This gap is due to the difficult appropriability of returns to investment by innovators: Arrow (1962) maintains that the primary output of R&D activities is knowledge, a non-rival good, whose spread cannot be avoided by the innovators; therefore, the returns from the acquisition of new knowledge cannot be completely appropriated by the firm undertaking the R&D investment with the result that firms under-invest. Moreover, the cost of financing R&D from sources outside the firm (Hall, 2002) may be higher because asymmetric information problems arise between borrowers and lenders. An additional gap between the private rate of return and the cost of financial capital may therefore exist with imperfect financial markets, entailing borrowing constraints for risky projects such as R&D.

The main strategic tools used by governments in almost all the OECD countries to stimulate innovation are public subsidies (in the form of concessional loans<sup>1</sup> or free grants or a mixture of the two) and tax incentives. Their purpose is to create an "additionality" effect: a subsidy reduces the cost of the R&D investment and so the firm receiving the subsidy performs additional projects whose overall expected profits would have been negative without the public intervention.

Although the evaluation of public policies is a widely recognized issue in all the OECD countries, there are few studies of Italian innovation policies, particularly at regional level: most regional laws do not oblige regional councils to prepare technical and financial reports on the measures carried out and there is a widespread lack of information.

This work would like to go some way towards filling this gap by shedding some light on whether regional management of the subsidies provided by the European Union (EU) Structural Funds has been effective. We focus on evaluating the effectiveness of two regional programmes of subsidies for the development and implementation of innovative technologies managed by the local government of a large Italian region. The first measure, which we call Measure A, is a concessional loan designed to encourage the

<sup>&</sup>lt;sup>\*</sup> Sections 1 and 2 are the joint work of Davide Fantino and Giusy Cannone. Davide Fantino is the author of all the data elaborations and the remaining sections of the text. We thank the regional financial company managing the subsidy programmes for providing us with the data on firms applying to the European Structural Funds. We thank three anonymous referees, Raffaello Bronzini, Bronwyn H. Hall, Silvia Magri, Guido Pellegrini, Massimo Sbracia, the participants to the 51<sup>st</sup> European Regional Science Association congress, to the 22<sup>nd</sup> annual scientific congress of the Associazione Italiana d'Ingegneria Gestionale and to several seminars in the Bank of Italy for their extremely useful comments. The usual disclaimers apply.

<sup>&</sup>lt;sup>1</sup> According to the IMF definition (IMF, 2003), concessional loans are loans extended on terms substantially more generous than market loans; they are usually provided with lower interest rates or longer repayment periods.

introduction of innovative plant, machinery and equipment, while the second, which we call Measure B, is a free grant to encourage general research activity by firms. It is worth examining the way policies of this type were designed because of their particular characteristics: their general aims and guidelines are defined by the EU, but the operative plans are developed at the national and regional level to take into account the local economic systems; the allocation of the available funds follows two sequential bargaining processes, one between EU and national governments and the other between national and local governments.

The data used in the policy evaluation exercise are taken from the official dataset of the regional financial agency managing the funds, merged with the balance sheets of the firms applying taken from the Cerved database. The methodology follows a difference-in-differences approach, where we compare the firms receiving the subsidy with those whose application was rejected; to minimize the bias due to the existing differences between the two groups we apply a matching procedure to the sample to select within the control group the firms that are most similar to the ones treated. The programmes have been effective in stimulating the targeted investments, even if the effects have been short-lived. Measure A had a positive impact on tangible investments during the project, but the effect faded within four years of the end of the project. Measure B had a significant positive impact on intangible investments only in the first year after the beginning of the project. When evaluating the effects of the two policy measures for some specific groups of firms, both measures seem to have been more effective for the smallest firms, while for the second measure this is also true for firms with a low credit rating. In the case of firms with a higher ratio between subsidy and total investments the effects are similar to the main ones for the first measure, while they are never significant for the second measure.

In general, the economic literature reveals little consensus on the ability of public funding to create additionality. Hussinger (2008) finds additionality, while evidence is mixed in Lach (2002), Almus and Czarnitzki (2003), and Duguet (2003); Wallsten (2000) find no effects. A previous study regarding some other EU subsidies can be found in Cariola et al. (2000): they evaluate the efficacy of two national programmes approved by the EU for small and medium firms (Multi-fund Operational Programmes and Community Initiative Programmes) and find a very weak effect due to the low average amount of funds provided and to some missing policy measures scheduled by the EU but not implemented. The results of our work for Measure B are consistent with those reported by Evangelista (2007), who uses data from the Eurostat Community Innovation Survey and finds a positive correlation between public funding and resources designated for innovation, but a very weak one between public funding and the introduction of new products. Bronzini and de Blasio (2006) investigate the effectiveness of Law 488/1992 in promoting tangible investments and find a temporal substitution between private and public investments, similarly to the results reported here for Measure A. The greater effectiveness of the subsidies for smaller firms is a common result in the literature (e.g. Lööf and Heshmati, 2007, for Swedish firms and Busom, 2000, for Spanish subsidies for R&D). Among the studies regarding regional policies, Bronzini and Iachini (2011) and Bronzini and

Piselli (2012) evaluate the effectiveness of the Regional Programme for Industrial Research, Innovation and Technological Transfer of the Emilia Romagna region (Regional Law 7/2002) and find that small firms increased their investment by about the amount of the subsidy, whereas there was no effect for larger firms. Gabriele et al. (2007) analyse the effect of local subsidies by the province of Trento (Provincial Law 4/1981) and find that subsidized firms increased their investments in the short run, with a transitory increase in labour productivity.

The rest of the paper is organized as follows: Section 2 briefly introduces the framework of the main EU policies drawn up in recent years to foster economic development, focusing on the policy analysed in this work; Section 3 describes the empirical strategy and the dataset; the results for the main sample, including robustness checks, are reported in Section 4, while Section 5 presents the analysis of the results obtained for four subsamples of firms; Section 6 concludes.

#### 2. Description of the policy

The local authorities have played an increasing role in the development of economic policies in recent years. The local management of the structural funds provided by the EU has provided regional governments with an important tool to stimulate innovative behaviour among firms and reduce the socio-economic gaps between different areas. The main instruments used by the EU to support these policy issues are the European Structural Funds.

The European Structural Funds are designed primarily to support regions with structural weaknesses. The main financial resources for the European Structural Funds policy come from the European Regional Development Fund, aimed at regulating regional unbalances, and the European Social Fund, which supports employment and social cohesion. The Structural Funds for the period 2000-2006 address three objectives: the structural adjustment of regions with lower economic and industrial development, in particular the areas whose GDP was less than 75 per cent of the EU average (Objective 1); the economic and social reconversion of areas with structural problems (Objective 2); and the general renewal of the education and employment systems (Objective 3). The first two objectives address the industrial system.

The framework for the management of the Structural Funds is laid down by the Council of the EU in Regulation 1260/1999. According to these guidelines the national governments approve the Strategic National Framework and the National Operative Programmes; negotiations take place between the EU and the national governments to determine the amount of funds assigned to each country. Subsequently, the regional governments define the Regional Operative Programmes in according to their national guidelines. A second negotiation takes place between the national government and the regions to establish the amount of funds available for each area. The regional government established its policy instruments and aims in a single regional programming document (DOCUP 2000-2006), approved by the European Commission.

The EU provided  $\notin$ 29 billion to the Italian government within the framework of the Structural Funds; around 70 per cent was allocated to Objective 1 and the remainder was divided equally between Objective 2 and Objective 3. The region we are considering is not classified as Objective 1, but had most of its municipalities classified as Objective 2 or as being phased-out, which means that they were authorized to receive subsidies in the period considered to consolidate the results obtained when they were Objective 2 areas in the previous period (1994-1999).

The regional programming document sets out the policy tools for Objective 2 local areas in four main fields of intervention: internationalization, support to the economic system, local development, and social cohesion. Under support to the economic system area, six types of subsidy are envisaged: incentives to innovative investments, to the creation of new firms, to the development of new infrastructures, to information and communication technologies, to tourism, and to stimulate research activity. We focus on the two policy measures to foster innovation activity: subsidies for innovative investments and subsidies for research activity.

The subsidies for innovative investments promote the introduction of innovative plant, machinery and equipment. They include the implementation of some national incentive plans (Laws 488/1992, 1329/1965, 598/1994 and 388/2000), an integration to grants already provided by the European Investment Bank, and a funding scheme managed directly by the region. We focus on the latter, that we call Measure A. The subsidies for research activity include two lines of intervention, both managed directly by the region: a general subsidy for firms' R&D activity (in the form of a non-refundable grant) and a specific contribution for environmental investments (in the form of a concessional loan). Only 28 loans have been awarded for the second intervention, not enough to allow a significant analysis; we therefore restrict our analysis to the other line of intervention, that we call Measure B.

The main characteristics of the two Measures are reported in Table 1. The funding scheme of Measure A is a partially public concessional loan for small and medium firms<sup>2</sup> operating in industry and services; applications were accepted between 10 December 2001 and 30 September 2007. The proposed project has to be completed within 18 months after selection, or 36 months for real-estate projects for tourism. The earliest funded project for this Measure was started on 10 December 2001 and the last one was completed on 15 November 2008; the earliest rejected project was supposed to begin on 10 December 2001, while the last one would have been completed on 10 June 2010. The funding scheme of the Measure is as follows: the public share of the funding is 50 per cent (with a maximum amount of half a million euros) and the borrowed amount has to be paid back in 48 months with no interest; a private bank provides the remaining funding at

<sup>&</sup>lt;sup>2</sup> According to the "Community Guidelines on State Aid for Small and Medium-Sized Enterprises" (Information from the European Commission C/213, published in 1996), small and medium firms are those with less than 250 employees, and with annual revenues lower than  $\notin$ 50 million or an annual balance-sheet value lower than  $\notin$ 43 million.

an interest rate equal to the 3-month Euribor, augmented by the market value of the 1-month spread in the repo market. The total amount of funding is paid to the firm after the end of the selection procedure. Projects are subsidized while funds are available in chronological order of approval; the other projects selected are put on a waiting list and funded when the firms already subsidized have paid back their loans.

Measure B is a free grant for small and medium firms operating in industry and services, except consortiums; the fund accepted applications between 29 January 2004 and 30 September 2006. The overall amount of expenditure of the projects must be not lower than  $\notin$ 25 thousand; projects have to be completed within 30 months of selection. The earliest funded project for this Measure was begun on 30 January 2004 and the last one was completed on 30 September 2008; the earliest project that was rejected was due to begin on 29 January 2004, while the last one would have been completed on 30 September 2008. The fund can pay up to 35 per cent of costs for pre-competitive research, patents or services and consulting for research activities and up to 50 per cent of feasibility studies. Moreover, the overall percentage contribution can be increased by up to 15 per cent if the project satisfies other criteria: 5 per cent if the investment is realized in an Article 83.7c area;<sup>3</sup> 10 per cent if there is a technology transfer between firms, countries or public and private sector; 15 per cent if the same project has been submitted to the EU R&D Framework Programme. However, the grant provided for each project cannot exceed 50 per cent of the total cost, with an upper bound of €150 thousand.

Under Measure A, the effective funding received by beneficiaries corresponds to the difference between the cost of financial capital at market price and the lower cost of financial capital at the price offered by the public agency. Considering a hypothetical average market interest rate of 3.5 per cent and public funding of  $\notin$ 500 thousand, the amount of saving corresponds to around  $\notin$ 18 thousand a year. The maximum subsidy received by beneficiaries for a 3-year project (the maximum allowed) is therefore about one-third of the maximum funding received directly by free grant for Measure B, that is  $\notin$ 150 thousand.

The agency in charge of managing these funds is a financial company controlled by the regional government. For both policy measures we analyse, a technical committee evaluates the formal aspects and the content of the applications. The evaluation is not comparative, but one-by-one, with priority of evaluation determined by the order of submission of the applications: if the formal requirements of the application are satisfied and the discretional evaluation of the project's merit is above a pre-defined numerical threshold, the subsidy is awarded. The discretional evaluation is based on the following general factors: 1. economic impact of the project on the firm, its workforce and its internationalization; 2. environmental sustainability; 3. incidence of female employment; 4. investment timing. Moreover, there is one Measure-specific criterion: 5. innovativeness for Measure A and compatibility with the EU Framework Programmes for R&D projects

<sup>&</sup>lt;sup>3</sup> Article 87 of the Treaty Establishing a Constitution for Europe regulates the procedure for state aid. In particular, Article 87.3a states that regional areas whose GDP is less than 75 per cent of the EU level are authorized to receive a larger amount of state aid; Article 83.7c lists the regional areas authorized to receive state aid, but for a lower amount.

for Measure B. Up to 5 points are assigned for each general factor and 1 additional point for the Measurespecific criterion; the pre-defined numerical threshold for the award is 6 for small firms and 9 for medium ones, while the theoretical maximum score is 21 points.

The total number of requests received for Measure A was 1837, of which 73 per cent (1339) were funded, for a total amount of public funds provided of around  $\notin$ 150 million. For Measure B, 1443 applications were submitted and 45 per cent (643) were funded, for a total amount of funds provided of around  $\notin$ 42 million.

### 3. Empirical strategy and data

#### 3.1 Empirical strategy

The evaluation of a policy programme serves to assess whether firms receiving the subsidy behave differently because of public intervention. This is a well-known counterfactual situation, because we cannot simultaneously observe the behaviour of a firm under the hypotheses of receiving a subsidy and not receiving it. Usually, this problem is econometrically solved by comparing the behaviour of the subsidized firms ("treated") with that of other firms ("controls") with similar characteristics not receiving the subsidy (Blundell and Costa Dias, 2000). A control group needs to be constructed and suitable techniques have to be used to conclude safely that it is a good proxy for the treated firms in the absence of the subsidy programme.

We use the set of the rejected firms as a pool of potential controls. Those firms should be similar to the treated ones because, by submitting their projects, they were self-selected as internally they considered themselves and their projects to be worthy of consideration for the subsidy; the bias due to the fact that the rejection of the project is a negative signal of efficiency should be minimized by the matching procedure for the observable characteristics and by the fixed effects included in the difference-in-differences methodology for the unobservables.

We use a nearest-neighbour matching technique to improve the similarity between the two groups: for each treated firm we choose a vector of characteristics and we select a firm among the controls whose characteristics minimize a distance objective function from the characteristics of the treated firm. The vector is chosen in the pre-treatment year to increase the likelihood of exogeneity with respect to the policy; it includes tangible and intangible investments, the size, economic, financial and profitability variables reported in Table 3. We use the ATECO 2007 code sector classification to identify six technology sectors: two of them are construction and services; the remaining four follow the OECD's partition of manufacturing by technology level (OECD, 2007; Hatzichronoglou, 1997). We partition the matching by technology sector.

The objective function we use is the numerical distance between the propensity score functions of the firms (Dehejia and Wahba, 2002; Frolich, 2004). The propensity score is calculated by using a probit model on the full sample of treated and controls to estimate the likelihood of a firm receiving the subsidy, given the vector of characteristics in the pre-treatment year, which therefore enter as covariates in the estimate. This type of matching procedure weights the effect of the policy on the outcome according to the distribution of the treated; we therefore estimate the average treatment effect on the treated (ATT). The matching procedure is done with replacement: a firm selected as a matching a treated firm is not removed from the pool of the potential matching of the other firms; the same control firm can therefore be selected more than once.

Given the different length of the project for different firms and the length of the interval in which firms were able to begin the project, the temporal overlapping between the timing of the projects was only partial and there were firms at different stages of project completion in a given year. We therefore decided to group firms according to the number of years after the beginning of the project and after its end: we observed each treated firm during the project after each year (up to three); we also observed it after completion for one and two years (and, for Measure A, for three and four years).<sup>4</sup> For each treated firm, we chose from our sample the available control firm whose characteristics in the pre-treatment year minimized the distance in the propensity scores and we associated the chosen control firm with the treated one.

The approach used to evaluate the impact of the public programme is the difference-in-differences technique. We compare the behaviour of the treated and the control firms between the pre-treatment period (2000) and a post-treatment year, where the treatment is the start of the subsidized project; we consider separately the effect during and after the project. The overall effect of the programme for a given year after either the beginning or the end of the project is calculated as the average of the differences in the outcome variable between each pair of treated and control firms, where the treated firm is observed in the required year after the treatment and the control is observed in the same calendar year as the associated treated:

$$\delta_{ij} = \frac{1}{N} * \sum \left( \Delta y_{ij}^T - \Delta y_{ij}^C \right) \tag{1}$$

where y is the outcome variable, the superscripts T and C are respectively for the treated and the control firms; the subscript i discriminates whether the treated in each pair of firms is observed during the project or after it, while the subscript j discriminates the number of years after the beginning or the end of the project; this pair of subscripts therefore determines exactly the state of the project of the treated firm in each pair.

The effect can be estimated in the following regression:

$$y_{ij} = \left[\alpha^T d^T + \alpha^C (1 - d^T)\right] + \left[\sum \beta_{ij} d_{ij}\right] + \left[d^T \sum \delta_{ij} d_{ij}\right] + \varepsilon$$
<sup>(2)</sup>

<sup>&</sup>lt;sup>4</sup> We considered a longer time window for Measure A because these projects began to be submitted about two years before those for Measure B (see Table 1).

where the Greek letters are the estimated coefficients,  $d^{T}$  is the dummy variable having value 1 when the firm belongs to the treated group and  $d_{ij}$  is the dummy variable having value 1 when the state of the project of the treated firm in each pair is exactly the one determined by *i* and *j* as defined before. In the equation there are therefore three terms: in the first square brackets there are fixed effects discriminating treated and control firms; in the second square brackets there are the effects discriminating the different years of the post-treatment period from the pre-treatment year; in the third square brackets are the effects discriminating the behaviour of the treated and the control firm for each year of the post-treatment period; it is this last group of coefficients that is interest to us and it corresponds to the coefficients defined in equation (1).

The hypothesis implied by this kind of approach is that the fixed effects discriminating treated and controls are sufficient to capture the differences in the unobservables, while the matching procedure takes care of the differences in the observables.

#### 3.2 Data

For the evaluation exercise we combine two datasets. The first one, provided by the regional agency managing the programs, includes information on the firms applying for the subsidy: name, date of application, programmed investments, assigned grants, revoked subsidies and renunciations. We merge these data with the balance sheets of the firms from the Cerved archives, which include information about virtually all Italian companies.

We used a balanced panel, including all firms with complete observations in the pre-treatment period (1999-2000) and the following years (2001-2008). To remove outliers, we polished the data, eliminating firms experiencing mergers, acquisitions and other similar operations; for each treatment year considered (2001-2008) and separately for funded and not funded firms, we also eliminated the first and last percentile of the variation in net total investments and revenues with respect to the pre-treatment year (2000).

Table 2 shows the distribution of firms by sector according to the OECD technology level classification, with the addition of two categories for the construction and services sectors. The first two columns refer to the firms applying respectively to Measure A and Measure B; the third column shows the sector distribution for firms in the region we are considering. The difference between the two Measures is clearly due to the particular characteristics of each. One half of the firms applying for Measure B come from high and medium-high technology sectors, while this share is one-fifth in the case of firms applying for Measure A and about 12 per cent when considering all the firms in the region.

The main outcome variables of our analysis are the net tangible and intangible investments measured after the treatment, scaled respectively by the tangible and intangible fixed assets observed in the pre-pre-treatment year (1999); net investments should be a good proxy to capture additional investments by the firm

after considering the substitution of depreciated fixed assets. The main drawback is that this variable may be influenced by the firm's fiscal policies.

In the analysis we also use several other aspects of firms' activity: size, economic structure, financial structure and profitability. The size of the firm is monitored through four variables: tangible fixed assets, intangible fixed assets, total assets, including both fixed and current assets, and sales revenue. Intangible fixed assets reveal whether the beneficiaries exploited R&D and patenting activities to make their firm grow. Tangible fixed assets give information about the physical components of the firm's accumulation process. Total assets and sales provide a more general picture and monitor the firm's overall level of activity.

Some aspects of the economic structure are taken into account by a labour productivity index, given by the ratio between value added and employees, and by net working capital, scaled either by sales or by total assets. These indices should monitor the firm's efficiency with respect to the amount of labour and financial resources used for its current management.

The financial structure is considered in order to monitor changes in the amount and characteristics of financial resources available to the firm. The debt structure is examined using the leverage index, that is the ratio between debts and the sum of debts and equity; we also analyse a modified version of leverage including only long-term financial debts in the numerator. Both these indices give signals about the firm's ability and necessity to access external financial resources in general and for the long term. Moreover, the average debt cost, given by the ratio between interest expenditure and total debt, should complement these indices with the perception of the risk associated with the debt and therefore indirectly with the reputation and perceived quality of the firm's assets. The cash-flow is used to monitor the firm's ability to create internal financial resources, normalized by either sales or total assets.

Finally, we observe three profitability indices: the return on equity (ROE), calculated as net income after taxes over equity, which monitors variations in the firm's ability to produce income for the shareholders; the return on assets (ROA), calculated as gross operating surplus over total assets, which is an index of the general profitability of the firm's assets; the return on investments (ROI), calculated as the gross operating surplus over the sum of equity and long term debts, which shows the profitability of the permanent component of the firm's financial resources.

After merging the datasets and removing outliers, the final size of our sample of subsidized firms is 212 for Measure A and 139 for Measure B. The descriptive statistics of the variables in the pre-treatment year (2000) are described in Table 3 for both Measure A and Measure B. There are relatively few significant differences between the two Measures. The most important are that the value of intangible fixed assets is

more than the double for Measure B with respect to Measure A. This dissimilarity clearly reflects the selfselection due to the different aims of the two Measures.

The matching procedure described in the previous section is then applied to the sample of treated firms and of the rejected ones to minimize the differences in the observables. We can test the quality of the final matched sample by checking the mean differences between treated and controls for the whole group of variables under observation in the pre-treatment. The results are reported in the first half of Table 4, where no mean difference is significant.

We also verify that the dynamics of size and investments of the treated and control firms are similar in the years immediately before the policy is implemented. If the treated firms have stronger dynamics than the control ones in the preceding years, the empirical detection of a positive impact of the policy could simply be due to different positions in the investment cycle. The results of the comparison are shown in the first half of Table 5: there is no evidence of differences in the average variations of size and investments of the two groups of firms between 1997 and 2000.

### 4. Empirical evidence

#### 4.1 Investment

The results obtained from the difference-in-differences estimation for both Measures, for tangible and intangible investments scaled by tangible and intangible fixed assets, are reported in Table 6. The errors are clustered by firm. Dummy variables for the different calendar years are also added to the specification of equation (2); their inclusion or exclusion does not affect the value of the coefficients of interest because they are common to both the treated and the control firm in each matched pair and are therefore differenced out in equation (1). Their inclusion should improve the precision of the estimated standard errors of the coefficients.

Even if the sample is balanced, the number of relevant firms for the estimation of each coefficient is not constant for two reasons: some projects lasted less than three years and were therefore observed only for the first and sometimes the second year; when considering the behaviour of the firms after the end of the projects, we were not able to observe the fourth, third, second or even first year after the end if the projects were completed respectively after 2004, 2005, 2006, or 2007, because our time window ended in 2008.

The table reports the estimates of the coefficients  $\delta_{ij}$  of equations (1) and (2), which are a measure of the effect of the policy. The first three columns show the impact during the project after one, two and three years; the fourth column shows the average effect during the project; the fifth, sixth, seventh and eighth

columns show the impact for each year after the end of the project and the ninth column shows the overall effect over the whole time window.

When considering Measure A, the effect on tangible investments during the project is positive and significant in the first year and on average. After completion, the impact becomes negative and is significant in the second year; the average effect over the whole time window is nil. When considering intangible investments, the effect is not significant.

When considering Measure B, we find a positive effect on intangible investments for the whole time window considered, but the impact is significant in the first year after the beginning of the project. The effect on tangible investments is never significant. When comparing the average effect on the intangible investments to intangible fixed assets ratio with the pre-treatment statistics shown, we find that the additionality effect of the policy is about twice the pre-treatment value.

#### 4.2 Robustness checks

The robustness of the results presented in the previous section has been confirmed by several checks with small modifications in the methodology and in the boundary of the sample.

In the first test we constructed an alternative balanced panel including all the firms with full data in the two pre-treatment years, in the start-year of the project and in the following two years. We now observe firms for a shorter period of time and we are therefore able to increase the number of those with the data requirements for inclusion in the balanced sample, improving the overall quality of the estimates. We repeat the analysis of the previous section for the three years after the beginning of the project, without distinguishing the firms observed during the project from those observed afterwards; in this way we reduce the composition effects due to the loss of firms with shorter projects or near the end of the time window for the estimation of each coefficient. The cost of this methodological change is that we are not able to discriminate different behaviours of the firms still working on their project and of those which have already completed it. The other aspects of the methodology are unchanged. The tests of the quality of the matching procedure for this alternative sample are shown in the second half of Tables 4 and 5 and we do not find significant mean differences in the variables monitored in the pre-treatment year.

The results for the impact of the subsidies are reported in Table 7. They largely confirm those reported in the previous section: for Measure A, we find positive effects in the first year, also significant for tangible investments, which become not significantly negative in the following years, when the share of completed projects increases; the overall effect of the policy over three years is still positive. For Measure B, we find positive effects for both types of investment in all the years; the coefficient is significant for intangible investments in the first year; the effects shown are slightly stronger than in Table 6.

In the second check we show the impact of the policy measures considering alternative investment ratios as objective variables. For both tangible and intangible investments we considered the scaling by the amount of sales and total assets. The results are shown in Table 8 and are in general qualitatively similar to those shown in Table 6. We find positive results during the project for both Measures and both types of investment when normalizing by total assets; the coefficients are significant for the tangible investments to total assets ratio for Measure A and for the intangible investments to total assets ratio for Measure B after the first year of the project; after completion we have mixed non-significant results, which reduce the average effect when compared with that obtained at the end of the project. When we scale investments by sales the results are similar, but never significant.

In the third test we consider different hypotheses of clustering for the errors of the estimated equation. In the first alternative we bootstrap by firm (1000 replications) the whole matching and difference-indifferences procedure to get a better estimate of the standard errors of the coefficients; the second alternative examines what happens if we assume clustering by economic sector (ATECO) instead of by firm; the third one assumes no clustering of the observations, i.e. the case of full independence among the errors. The results for all the cases are reported in Table 9; changing the hypothesis of clustering of the equation errors does not substantially modify the conclusions of the analysis.

In the fourth check we consider whether the addition of other control variables would change our conclusions. With the aim of isolating the direct effects of the policy from those coming from the evolution of other aspects of the firm, we add to the specification of equation (2) the following variables, observed in the same year as the outcome variable, covering the main aspects of the firm's activity: the amount of revenue from sales, the net working capital to sales ratio, the cash-flow to sales ratio, the leverage index, the ratio between long-term financial debts and the sum of equity and debts, the average cost of the debt and the return on assets profitability index. The results are reported in Table 10 and are qualitatively similar to the main case: during the project we find positive effects, significant for the first year for tangible investments for Measure A and for intangible investments for Measure B; after the end of the project, for Measure A the effect is almost always negative for both types of investment and is significant in the second year.

#### 4.3 Other outcome variables

In addition to the effect on investment, in this section we consider whether the policy influences other aspects of the firm's activity. The results regarding size, economic structure, financial structure and profitability for Measure A are reported in Table 11a.

The variables accounting for the firm's size show a positive impact during the project and immediately after. The coefficients are significant for sales in the three years following the end of the project. The effect

on intangible fixed assets is weaker and sometimes becomes negative. The average impact of the policy is always positive, but never significant. When considering the economic structure of the firm, net working capital is not consistently significant in any estimation, showing that there is no relevant effect on the management of the firm's current activity. The same is true when considering the value added per employee, showing that the additional investments are not more productive than the old ones.

The financial structure of the firm is partially affected by the policy. The cash-flow indices are usually positive, but not significant, during the project and they become not significantly negative from the second year after the end of the project: even if the subsidy provides additional liquidity, the subsidized projects do not seem to be more efficient in generating additional current financial resources than the total assets already accumulated. While the leverage is not significantly influenced by the policy, the effect on long-term financial debts is positive up to three years after the end of the project and significant in the second year and on average during the project; the firm uses the subsidy, whose amount is included in financial long-term debt, to substitute short-term debt. The average cost of financial capital shows a significant decrease of 0.8 percentage points on average during the project and of 0.5 over the whole time window, probably associated with the concessional nature of the loan.

Finally, the profitability indices do not seem to be significantly and consistently affected by the public programme during the development of the project or after its completion, either in terms of return on equity, return on assets or return on investment. Given the positive dynamics of total assets and sales, this is a signal that the additional assets accumulated with the subsidized project have the same profitability as those accumulated previously.

The impact on the other outcome variables for Measure B is shown in Table 11b. The impact on firm size is never significant; it is slightly positive, for the two types of fixed asset, while the effect on sales and total assets is mixed. The impact on net working capital is not significantly negative, while that on value added per employee is not significantly positive. Moreover, the financial structure is not heavily affected: the impact on cash-flow is weak and almost always negative, showing that the project is not creating additional financial resources in the time window considered and there are indications that it is still draining them. The debt structure is substantially unchanged and there are no relevant effects on the average cost of debt; this difference in the effectiveness with respect to the other Measure may be due to the fact that this subsidy is a free grant, while the other Measure is a concessional loan. The results concerning profitability are generally weak and not significant.

#### 5. Effect on some subsamples

We now evaluate the impact of the Measures on four sub-samples in order to verify whether particular conditions could influence the efficacy of the programme. The matching procedure is repeated for each of

them. For all the sub-samples, the usual mean comparisons for the pre-treatment year after the matching procedure, useful to test its quality, are reported in Table 12, while the results of the difference-in-differences estimation are shown in Table 13a for Measure A and in Table 13b for Measure B.

The first sub-sample includes the firms whose credit rating in the Cerved dataset is particularly low; we include in the analysis the firms from the three lowest classes of rating, which are those defined as risky in the Cerved balance-sheet analysis. These firms are more likely to be financially constrained and may benefit from the additional financial capital provided by the subsidy. We find that for Measure B the impact of the policy on intangible investments is stronger than in the main case, while for Measure A the policy has somewhat weaker effects on this class of firms.

In the second and third sub-samples we split the overall sample according to dimension. In one case we include the smallest firms, defined as those with an amount of sales under the median of the distribution in the pre-treatment year; in the other we include all the remaining firms. Small firms had an advantage in the selection process because the threshold of acceptance was lower for them. This is because they have a peculiar role in the economic structure of the region and more in general of Italy; moreover, they usually have additional difficulties in accessing financial markets and therefore may get more benefit from the subsidy (Carmignani and D'Ignazio, 2011; Hussinger, 2008). We find that for both Measures the impact of the policy on the investment ratios of the smallest firms is stronger than for the other firms during the projects, while after their end the coefficients are never significant and for Measure A the effect fades once again. The effect for the largest firms, which is still substantial for Measure A, becomes not significant for Measure B.

The fourth sub-sample includes the subsidized firms for which the ratio between the amount of subsidy received and the total net investments is above the median of the distribution of the total sample of subsidized firms, to understand whether a high intensity of subsidy may improve the efficacy of the programme. Indeed, if the subsidy covers a small share of the investments, its influence on the firm's behaviour may be not observable because it is hidden by the overall variance of the investments. The effects on the net investments to fixed assets ratio are similar to those reported for the main sample, even if they are never significant for Measure B; the estimated impact is on average slightly greater.

#### 6. Conclusions

Despite the large number of studies on the evaluation of R&D public subsidies, there is still controversy about their effectiveness. The aim of this paper is to contribute to this debate by providing some evidence of the impact of two policy measures that use the European Structural Funds but are locally managed, which aim to support innovative activity for small and medium firms in a large Italian region.

The results obtained, which are robust to several changes in methodology, show that the programmes are effective in stimulating the targeted investments, even if the effects are short-lived. In the case of Measure A, which is designed to stimulate the introduction of innovative plant, machinery and equipment through a public concessional loan, the subsidy has a positive impact on tangible investment during the project, but the effect fades within four years of its completion. The effectiveness of the subsidy therefore seems weak, providing a temporary boost to the firm's activity.

Results obtained from Measure B, which promotes research projects through a free grant, show a positive impact on intangible investment, coherently with the Measure's focus on research and knowledge capital; even if the effect is significant only in the first year after the beginning of the project, this policy seems to be at least partially effective.

When evaluating the effects of the two policy measures for some specific groups of firms, the effects are similar to the main ones for Measure A, while are never significant for Measure B for firms with a higher ratio of subsidy to total investments. Both measures seem to be more effective for the smallest firms, while for the second measure this is also true for firms with a low credit rating. In designing policy measures of this type, focusing on these specific groups of firms may help to increase their effectiveness.

The different degree of effectiveness of the two subsidies can be explained by looking at the projects targeted. Projects financed by Measure A are at the end of their R&D cycle, when the risk associated with the project is low: basic research has been completed, firms are nearing the end of the development stage and they have an industrial plan for the introduction of the innovation in the production process; firms are therefore already committed to their projects because they have spent a lot of time and resources in the previous stages of creating the innovation and the likelihood of financing a project that will not be completed is very small. The subsidy therefore modifies the firm's schedule in such a way that the timing of investments corresponds to that of the Measure; moreover, the award of the subsidy simplifies the search for financiers on the financial markets and so may reduce the time needed to complete the project.

Measure B is a general R&D subsidy, which can therefore be used by the firm to finance the earliest and riskiest stages of the research activity; since the subsidy is a free grant covering a large share of the costs, its award allows the firm to pass a large part of the financial consequences of failure onto the public sector. The likelihood of creating additionality is therefore quite high and the firm has an incentive to apply for projects with low social value added; in this case the effectiveness of the selection process is particularly important to ensure that the projects financed are of public interest and that they are able not only to create general knowledge, but also to be effectively applied in the production process.

## Tables

# Table 1: Summary of the characteristics of the Measures

	Measure A	Measure B
Number of applications	1837	1443
Number of firms funded (percentage)	1339 (73%)	643 (45%)
Overall amount of subsidy granted	150 mln €	42 mln €
Earliest start date of a funded project	10/12/2001	30/01/2004
Earliest end-date of a funded project	31/12/2001	30/06/2004
Last start-date of a funded project	15/05/2008	01/03/2007
Last end-date of a funded project	15/11/2008	30/09/2008
Earliest theoretical start-date of a rejected project	10/10/2001	29/01/2004
Earliest theoretical end-date of a rejected project	10/03/2002	15/05/2004
Last theoretical start-date of a rejected project	01/07/2008	01/03/2007
Last theoretical end-date of a rejected project	10/06/2010	30/09/2008
Average length of the funded projects	330 days	604 days
Average length of the rejected projects	309 days	520 days

# Table 2: Firms obtaining the subsidy (by technology sectors) and sector distribution of all firms in the region in 2000

Technology sector	Measure A	Measure B	Region
High technology (OECD)	2.75%	11.06%	2.22%
Medium-high technology (OECD)	17.56%	39.38%	9.95%
Medium technology (OECD)	34.06%	16.52%	11.41%
Low technology (OECD)	14.01%	6.05%	11.47%
Construction	9.50%	2.51%	10.83%
Services	22.13%	24.49%	54.12%

Variable	Measure	Mean	Std. Dev.	Min	Median	Max
		Size				
Intangible fixed assets	А	61.21	93.33	0	27	517
Intaligible lixed assets	В	124.10	371.84	0	33	3918
Tangible fixed assets	A	745.06	1061.83	2	409	6915
	В	825.06	1107.73	6	347	5363
Sales	A	3980.68	4239.55	85	2711	40645
Cales	В	4720.66	4473.58	137	3391	24297
Total assets	A	3292.63	3667.70	70	2062	34730
	В	4113.48	4054.33	92	3058	24296
		vestment				
Intangible investments	A	2.50	32.48	-123	-1	194
-	В	14.24	73.98	-124	0	402
Intangible investments over	A	0.0014	0.0226	-0.1294	-0.0005	0.1794
sales	В	0.0063	0.0444	-0.0927	0	0.4233
Intangible investments over	A	0.0053	0.0512	-0.1310	-0.0006	0.6316
total assets	B	0.0074	0.0399	-0.0687	0	0.2761
Intangible investments over	A	0.7402	4.2565	-1	-0.0935	48
intangible fixed assets	В	1.0812	6.2936	-1	0	71
Tangible investments	A	83.54	264.60	-324	11.5	1787
•	B	73.85	302.64	-1216	4	2158
Tangible investments over	A	0.0452	0.1630	-0.1134	0.0057	1.2718
sales	B	0.0434	0.1620	-0.1123	0.0013	1.2718
Tangible investments over	A	0.0601	0.2762	-0.1257	0.0079	3.1579
total assets	B	0.0471	0.1894	-0.1693	0.0016	1.9346
Tangible investments over	A	0.2482	1.1229	-0.6000	0.0556	13.6866
tangible fixed assets	B	0.3332	1.3458	-0.7901	0.0182	13.6866
		omic stru		0.0054	0.0450	0.0445
Value added on sales	A	0.3324	0.1526	0.0051	0.3150	0.9115
Not working conital over	B	0.3306	0.1378	0.0601	0.3150	0.8579
Net working capital over	A	-0.0832	0.3695	-3.2115	-0.0288	0.7639
sales	B A	-0.0898 -0.0718	0.2803 0.2849	-1.5646 -1.0677	-0.0431 -0.0441	0.8180 0.5746
Net working capital over total assets	B	-0.0718	0.2649	-0.8763	-0.0441	0.3748
855615		ncial struc		-0.0703	-0.0557	0.4024
	А	0.0607	0.0597	-0.1920	0.0454	0.3124
Cash-flow over sales	B	0.0609	0.0570	-0.0478	0.0434	0.3013
	A	0.0003	0.0616	-0.1097	0.0581	0.3059
Cash-flow over total assets	-	0.0724	0.0528	-0.0616	0.0588	0.2423
	В А	0.7684	0.0328	0.1739	0.8102	1.0504
Leverage	B	0.7824	0.1585	0.3380	0.8361	0.9859
Long-term financial debts	A	0.1024	0.1244	0.0000	0.0612	0.7731
over (debts+equity)	B	0.0853	0.1197	0	0.0517	0.6648
· · · · · ·	A	0.0325	0.0226	0	0.0280	0.1372
Average cost of debt	В	0.0334	0.0220	0	0.0200	0.1713
		rofitability		0	0.0270	0.1710
	A	0.2689	0.7987	-6.6667	0.1604	5.4074
ROE	В	0.2986	0.4001	-0.9885	0.1927	2.9487
	A	0.2300	0.0745	-0.0827	0.0633	0.4349
ROA	В	0.0776	0.0624	-0.0702	0.0668	0.3121
	A	0.1621	0.2510	-1.0556	0.1224	1.9091
ROI	В	0.2122	0.2277	-0.1583	0.1418	1.5532
	_			0000		

Table 3: Descriptive statistics of treated firms in the pre-treatment period (2000)

The Cerved dataset includes 212 treated firms for Measure A and 139 firms for Measure B. The size variables and the raw level of investments are measured in thousand euros. The denominators of the investments ratios are measured in the pre-pre-treatment year (1999).

Table 4: Mean differences in firm characteristics in the pre-treatment period (2000) after matching for the full window sample and reduced window sample

		indow		l window		
	Measure	Measure	Measure	Measure		
	A	В	A	В		
Intangible fixed assets	-10.854	-30.770	-14.434	-32.962		
	(0.163)	(0.356)	(0.155)	(0.116)		
Tangible fixed assets	-17.363	90.252	79.378	-81.694		
Tangible fixed assets	(0.834)	(0.487)	(0.321)	(0.403)		
Sales	-46.425	644.094	570.707	-457.849		
Gales	(0.898)	(0.278)	(0.404)	(0.298)		
Total assets	-30.392	386.568	350.108	-352.946		
10181 833613	(0.917)	(0.418)	(0.213)	(0.279)		
Intangible investments	3.882	-6.568	-4.341	-12.522		
	(0.277)	(0.413)	(0.148)	(0.119)		
Intangible investments over	0.002	0.010	0.003	0.007		
sales	(0.330)	(0.184)	(0.388)	(0.371)		
Intangible investments over	-0.002	0.001	0.003	-0.001		
total assets	(0.517)	(0.768)	(0.420)	(0.902)		
Intangible investments over	-0.319	0.120	0.211	0.669		
intangible fixed assets	(0.332)	(0.853)	(0.734)	(0.328)		
Tangible investments	5.788	30.475	23.361	37.688		
-	(0.795)	(0.357)	(0.339)	(0.217)		
Tangible investments over	-0.012	-0.013	-0.014	0.004		
sales	(0.471)	(0.409)	(0.213)	(0.769)		
Tangible investments over	-0.015	-0.013	-0.015	0.027		
total assets	(0.377)	(0.461)	(0.217)	(0.150)		
Tangible investments over	-0.132	-0.120	-0.133	0.055		
tangible fixed assets	(0.298)	(0.337)	(0.290)	(0.635)		
Value added per employee	0.018	-0.009	0.021	-0.024		
value added per employee	(0.259)	(0.627)	(0.100)	(0.130)		
Net working capital over sales	0.015	-0.022	0.019	0.005		
	(0.638)	(0.507)	(0.486)	(0.878)		
Net working capital over total	0.028	-0.022	0.006	-0.019		
assets	(0.264)	(0.508)	(0.786)	(0.501)		
Cash-flow over sales	-0.004	0.007	-0.005	-0.004		
	(0.154)	(0.457)	(0.177)	(0.529)		
Cash-flow over total assets	-0.003	-0.004	-0.006	0.000		
	(0.211)	(0.615)	(0.123)	(0.994)		
Leverage	0.031	0.005	0.006	0.025		
	(0.250)	(0.815)	(0.686)	(0.145)		
Long-term financial debts	0.035	0.022	0.001	0.013		
over (equity+debts)	(0.300)	(0.131)	(0.953)	(0.280)		
Average cost of debt	-0.006	-0.005	-0.002	-0.002		
· · · · · · · · · · · · · · · · · · ·	(0.100)	(0.157)	(0.232)	(0.327)		
ROE	0.162	-0.261	0.168	-0.090		
	(0.292)	(0.198)	(0.231)	(0.203)		
ROA	-0.010	-0.016	-0.006	-0.010		
	(0.200)	(0.131)	(0.205)	(0.123)		
ROI	0.049	-0.032	0.106	-0.079		
	(0.266)	(0.301)	(0.400)	(0.267)		
# observations	424	278	498	372		

The size variables and the raw level of investments are measured in thousand euros. Significance level reported between brackets.

The denominators of the investments ratios are measured in the pre-pre-treatment year (1999). Full window sample: 1999-2008.

Reduced window sample: 1999-2000 and the 3 years immediately following treatment.

Table 5: Mean differences in the dynamics of the main variables before the beginning of the policy measures (1997-2000) for the full window and reduced window sample

	Full w Measure A	indow Measure B	Reduced Measure A	l window Measure B
Variation of intangible fixed	-1.254	-9.107	0.737	-11.781
assets	(0.617)	(0.182)	(0.830)	(0.108)
Variation of tangible fixed	28.269	-9.843	33.088	7.321
assets	(0.122)	(0.584)	(0.102)	(0.648)
Variation of total assets	72.897	32.168	57.008	-23.397
	(0.124)	(0.589)	(0.200)	(0.638)
Variation of sales	58.019	-18.214	65.855	-82.952
variation of sales	(0.219)	(0.849)	(0.300)	(0.227)
Variation of intangible	-2.808	-3.490	-10.574	-2.116
investments	(0.494)	(0.690)	(0.134)	(0.770)
Variation of tangible	-0.334	-20.203	7.418	18.063
investments	(0.987)	(0.184)	(0.674)	(0.500)
# observations	424	278	498	372

Thousand euros.

Significance level reported between brackets.

Full window sample: 1999-2008.

Reduced window sample: 1999-2000 and the 3 years immediately following the treatment.

## Table 6: Difference-in-differences results during the project and after: net investments

	Years during the project		Average during the Years after the project			Overall average			
	1	2	3	project	1	2	3	4	average
			Meas	ure A					
Intangible investments over	0.64	1.23	1.78	1.01	0.22	-1.57	-0.73	0.72	0.25
intangible fixed assets	(1.17)	(1.15)	(2.52)	(0.83)	(2.00)	(0.96)	(1.31)	(1.38)	(0.71)
Tangible investments over	0.48***	0.10	0.38	0.33**	-0.25	-0.52**	-0.05	-0.21	0.00
tangible fixed assets	(0.18)	(0.25)	(0.40)	(0.16)	(0.25)	(0.25)	(0.26)	(0.43)	(0.15)
# observations	424	324	114	862	300	280	230	180	1852
	-		Meas	ure B	-				-
Intangible investments over	2.94*	1.97	1.28	2.18	1.46	1.46			1.92
intangible fixed assets	(1.53)	(2.14)	(2.60)	(1.59)	(2.40)	(2.58)			(1.22)
Tangible investments over	0.16	0.27	0.24	0.22	-1.14	0.80			0.03
tangible fixed assets	(0.25)	(0.36)	(0.42)	(0.23)	(1.74)	(0.56)			(0.38)
# observations	278	276	170	724	230	174			1128

Clustered standard errors by firm included between brackets.

The denominators of the investments ratios are measured in the pre-pre-treatment year (1999).

\*,\*\*,\*\*\* for significance levels of the coefficients respectively of 90%, 95% and 99%.

	Years after t	he beginning c	of the project	Ι.
	1	2	3	Average
	Measu	re A		
Intangible investments over	1.77	-1.06	-0.71	0.06
intangible fixed assets	(1.98)	(2.37)	(2.83)	(2.10)
Tangible investments over	0.42**	-0.05	-0.07	0.11
tangible fixed assets	(0.19)	(0.20)	(0.20)	(0.15)
# observations	498	470	418	1386
	Measu	re B		-
Intangible investments over	4.06**	2.36	1.99	2.80
intangible fixed assets	(1.81)	(5.77)	(1.78)	(2.25)
Tangible investments over	0.55	0.33	0.16	0.34
tangible fixed assets	(0.37)	(0.32)	(0.36)	(0.27)
# observations	372	372	370	1114

## Table 7: Difference-in-differences results: reduced window

Clustered standard errors by firm included between brackets.

The denominators of the investments ratios are measured in the pre-pre-treatment year (1999).

\*,\*\*,\*\*\* for significance levels of the coefficients respectively of 90%, 95% and 99%.

#### Table 8: Difference-in-differences results during the project and after: other investment measures

	Years o	during the	e project	Average during the	Years after the project			Overall	
	1	2	3	project	1	2	3	4	average
Measure A									
Intangible investments	0.003	0.011	0.010	0.007	0.000	-0.008	-0.007	0.002	0.002
over sales	(0.008)	(0.010)	(0.014)	(0.007)	(0.008)	(0.007)	(0.010)	(0.015)	(0.006)
Intangible investments	0.000	0.017	0.005	0.007	-0.008	-0.011	-0.009	0.000	-0.001
over total assets	(0.009)	(0.014)	(0.019)	(0.009)	(0.010)	(0.008)	(0.010)	(0.016)	(0.006)
Tangible investments	0.032	0.022	0.022	0.027	-0.034	-0.030	0.019	-0.043	0.001
over sales	(0.027)	(0.041)	(0.036)	(0.024)	(0.043)	(0.036)	(0.058)	(0.103)	(0.032)
Tangible investments	0.051*	0.003	0.046	0.032	-0.050	-0.044	0.002	-0.024	-0.002
over total assets	(0.027)	(0.043)	(0.097)	(0.026)	(0.047)	(0.037)	(0.057)	(0.092)	(0.032)
# observations	424	324	114	862	300	280	230	180	1852
	-		Me	easure B	_				-
Intangible investments	0.027	0.016	0.009	0.018	0.005	0.026			0.017
over sales	(0.017)	(0.031)	(0.015)	(0.011)	(0.017)	(0.018)			(0.018)
Intangible investments	0.020*	0.011	0.008	0.014	-0.005	0.035			0.013
over total assets	(0.011)	(0.017)	(0.019)	(0.010)	(0.018)	(0.025)			(0.009)
Tangible investments	0.010	0.011	0.029	0.015	-0.124	0.044			-0.009
over sales	(0.028)	(0.036)	(0.038)	(0.027)	(0.081)	(0.038)			(0.029)
Tangible investments	0.007	0.016	0.018	0.013	-0.109	0.069			-0.003
over total assets	(0.029)	(0.038)	(0.045)	(0.029)	(0.077)	(0.045)			(0.025)
# observations	278	276	170	724	230	174			1128

Clustered standard errors by firm included between brackets.

The denominators of the investments ratios are measured in the pre-pre-treatment year (1999).

\*,\*\*,\*\*\* for significance levels of the coefficients respectively of 90%, 95% and 99%.

Table 9: Difference-in-differences results during the project and after: robustness checks on
standard errors

	Years during the project		Average during the	Years after the project		Overall			
	1	2	3	project	1	2	3	4	average
	Measure A								
Intangible	0.64	1.23	1.78	1.01	0.22	-1.57	-0.73	0.72	0.25
investments	(1.16)	(1.11)	(2.48)	(0.85)	(1.97)	(1.02)	(1.37)	(1.44)	(0.70)
over intangible	[1.19]	[1.29]	[2.15]	[0.69]	[2.06]	[1.01]	[1.31]	[1.45]	[0.66]
fixed assets	{1.03}	{1.18}	{1.99}	{0.72}	{1.23}	{1.27}	{1.40}	{1.59}	{0.49}
Tangible	0.48	0.10	0.38	0.33	-0.25	-0.52	-0.05	-0.21	0.00
investments	(0.18)***	(0.24)	(0.39)	(0.16)*	(0.26)	(0.25)**	(0.26)	(0.46)	(0.15)
over tangible	[0.13]***	[0.22]	[0.44]	[0.12]***	[0.33]	[0.23]**	[0.28]	[0.51]	[0.14]
fixed assets	{0.18}***	{0.20}	{0.34}	{0.12}***	{0.21}	{0.22}**	<sup>*</sup> {0.24}	{0.27}	{0.08}
# observations	424	324	114	862	300	280	230	180	1852
	•			Measure	В				
Intangible	2.94	1.97	1.28	2.18	1.46	1.46			1.92
investments	(1.94)	(2.27)	(2.70)	(1.61)	(2.53)	(2.69)			(1.20)
over intangible	[1.47]**	[1.64]	[2.34]	[1.29]*	[2.92]	[3.30]			[1.39]
fixed assets	{1.82}	{1.82}	{2.32}	{1.13}*	{2.00}	{2.30}			<b>{1.10}</b> *
Tangible	0.16	0.27	0.24	0.22	-1.14	0.80			0.03
investments	(0.25)	(0.34)	(0.41)	(0.22)	(1.83)	(0.56)			(0.37)
over tangible	[0.12]	[0.37]	[0.23]	[0.16]	[1.94]	[0.51]			[0.41]
fixed assets	{0.56}	{0.57}	{0.73}	{0.35}	<b>{</b> 0.62}	{0.72}			{0.28}
# observations	278	276	170	724	230	174			1128

Boostrapped clustered standard errors by firm (1000 replications) included between round brackets. Clustered standard errors by ATECO sector included betweens square brackets.

Non-clustered standard errors included betweens braces.

The denominators of the investments ratios are measured in the pre-pre-treatment year (1999). \*,\*\*,\*\*\* for significance levels of the coefficients respectively of 90, 95 and 99% when using the reported standard errors.

Table 10: Difference-in-differences results (with control variables) during the project and after:
net investments

	Years during the project			Average Years after the project during the					Overall	
	1	2	3	project	1	2	3	4	average	
Measure A										
Intangible investments	0.68	1.05	1.73	0.96	0.03	-1.69*	-0.93	0.85	0.16	
over intangible capital	(1.19)	(1.12)	(2.53)	(0.88)	(2.08)	(0.98)	(1.34)	(1.49)	(0.74)	
Tangible investments	0.47***	0.13	0.36	0.32**	-0.24	-0.46**	0.16	-0.16	0.05	
over tangible capital	(0.18)	(0.25)	(0.38)	(0.15)	(0.26)	(0.20)	(0.29)	(0.44)	(0.16)	
# observations	424	324	114	862	300	280	230	180	1852	
			Ме	asure B	-				-	
Intangible investments	2.67*	1.81	1.62	2.10	1.16	3.05			2.05	
over intangible capital	(1.58)	(2.21)	(2.57)	(1.60)	(2.55)	(2.72)			(1.26)	
Tangible investments	0.12	0.31	0.24	0.22	-0.15	0.57			0.20	
over tangible capital	(0.25)	(0.35)	(0.43)	(0.23)	(1.71)	(0.37)			(0.35)	
# observations	278	276	170	724	230	174			1128	

Clustered standard errors by firm included between brackets.

The denominators of the investments ratios are measured in the pre-pre-treatment year (1999). \*,\*\*,\*\*\* for significance levels of the coefficients respectively of 90%, 95% and 99%.

	Years during the project			Average Years after the project					Overall
	1	2	3	during the project	1	2	3	4	average
Sales	411.3	66.9	448.0	581.78	851.8*	876.1*	1011.7*	537.1	838.65
Sales	(314.3)	(539.9)	(538.2)	(415.88)	(478.7)	(530.6)	(564.6)	(819.8)	(542.90)
Total assets	273.0	638.9	677.0	464.02	408.0	312.3	697.3	580.5	472.33
Total assets	(436.6)	(825.2)	(468.9)	(555.17)	(526.2)	(699.7)	(559.0)	(610.5)	(558.84)
Intangible fixed assets	-1.78	-1.59	11.42	0.03	-9.37	-2.82	2.92	51.70	3.46
Intaligible lixed assets	(29.36)	(93.04)	(52.27)	(50.09)	(70.69)	(78.86)	(54.49)	(64.60)	(54.87)
Tangible fixed assets	89.60	98.77	131.18	98.55	47.84	35.13	36.20	36.32	66.95
Tangible fixed assets	(88.08)	(115.44)	(166.44)	(94.98)	(119.56)	(151.16)	(166.97)	(230.70)	(108.60)
Net working capital over	-0.040	0.002	-0.080	-0.030	0.066	-0.013	0.136	-0.027	0.009
sales	(0.079)	(0.083)	(0.058)	(0.073)	(0.098)	(0.092)	(0.176)	(0.142)	(0.091)
Net working capital over	-0.040	0.005	-0.046	-0.024	0.021	-0.021	-0.004	-0.089	-0.020
total assets	(0.068)	(0.065)	(0.044)	(0.060)	(0.077)	(0.074)	(0.105)	(0.092)	(0.068)
Value added per	0.015	0.013	0.023	0.015	0.000	-0.007	-0.014	-0.007	0.004
employee	(0.020)	(0.022)	(0.034)	(0.019)	(0.022)	(0.027)	(0.028)	(0.034)	(0.019)
Cash-flow over sales	0.010	0.004	0.006	0.007	0.007	-0.043	-0.072	-0.073	-0.018
	(0.013)	(0.014)	(0.011)	(0.011)	(0.012)	(0.031)	(0.053)	(0.437)	(0.018)
Cash-flow over total	0.016	0.016	0.000	0.014	0.022	-0.001	-0.010	-0.044	0.004
assets	(0.010)	(0.013)	(0.012)	(0.010)	(0.015)	(0.013)	(0.014)	(0.034)	(0.009)
Leverage	0.043	0.012	0.051	0.033	0.008	0.011	0.036	0.040	0.026
-	(0.045)	(0.037)	(0.034)	(0.037)	(0.040)	(0.046)	(0.048)	(0.048)	(0.038)
Long-term financial	0.048	0.063**	0.048	0.054*	0.060	0.036	0.066	-0.019	0.047
debts over (equity+debts)	(0.033)	(0.029)	(0.042)	(0.030)	(0.038)	(0.046)	(0.046)	(0.062)	(0.034)
Average cost of debt	-0.009***	-0.007**	-0.006*	-0.008***	-0.006*	-0.004	-0.001	0.000	-0.005*
Average cost of debt	(0.003)	(0.004)	(0.003)	(0.003)	(0.003)	(0.003)	(0.005)	(0.004)	(0.003)
ROE	0.27	0.26	0.53	0.32	0.46	0.24	0.36	0.26	0.34
ROE	(0.46)	(0.34)	(0.47)	(0.62)	(0.40)	(0.42)	(0.48)	(0.47)	(0.63)
ROA	-0.01	-0.02	-0.01	-0.02	-0.01	-0.02	-0.02	-0.03	-0.02
ROA	(0.02)	(0.02)	(0.02)	(0.01)	(0.01)	(0.02)	(0.01)	(0.02)	(0.02)
ROI	0.40	0.39	0.21	0.45	1.06	0.32	0.19	0.31	0.51
	(0.30)	(0.33)	(0.23)	(0.40)	(0.94)	(0.75)	(0.23)	(0.24)	(0.51)
# observations	424	324	114	1852	300	280	230	180	990

## Table 11a: Difference-in-differences results for Measure A during the project and after: other variables

The size variables and the raw level of investments are measured in thousand euros.

Clustered standard errors by firm included between brackets.

\*,\*\*,\*\*\* for significance levels respectively of 90%, 95% and 99%.

Table 11b: Difference-in-differences results for Measure B during the project and after: other variables

	Years	during the	project	Average during the	Years a pro	Overall	
	1	2	3	project	1	2	average
Sales	-664.8	-260.1	51.2	-153.50	70.8	335.9	-153.50
Sales	(761.2)	(508.1)	(452.3)	(467.88)	(483.0)	(634.1)	(467.88)
Total assets	-58.9	514.8	677.5	389.96	453.3	544.3	389.96
Total assets	(510.6)	(312.0)	(418.4)	(299.93)	(435.1)	(634.1)	(299.93)
Intangible fixed assets	42.83	13.86	20.91	26.64	39.19	54.31	33.47
Intangible lixed assets	(43.37)	(54.69)	(44.94)	(40.77)	(68.01)	(89.06)	(47.91)
Tangible fixed assets	119.89	293.30	104.22	182.32	57.83	167.21	154.61
Tangible lixed assets	(121.00)	(217.82)	(193.37)	(140.01)	(211.32)	(276.84)	(159.70)
Net working capital over	-0.015	-0.044	-0.015	-0.026	-0.027	-0.094	-0.092
sales	(0.037)	(0.052)	(0.050)	(0.034)	(0.055)	(0.069)	(0.100)
Net working capital over	0.014	0.029	0.030	0.024	0.038	0.033	-0.037
total assets	(0.017)	(0.029)	(0.020)	(0.019)	(0.029)	(0.039)	(0.036)
Value added per	-0.004	0.004	0.003	0.001	0.004	-0.005	0.028
employee	(0.018)	(0.016)	(0.013)	(0.015)	(0.028)	(0.033)	(0.022)
Cash-flow over sales	-0.029	-0.018	-0.009	-0.020	-0.002	-0.017	0.000
Cash-now over sales	(0.018)	(0.018)	(0.011)	(0.015)	(0.013)	(0.016)	(0.017)
Cash-flow over total	0.007	0.001	0.026	0.009	0.028	0.031	-0.016
assets	(0.022)	(0.034)	(0.027)	(0.024)	(0.032)	(0.029)	(0.012)
	0.042	0.020	0.007	0.025	-0.001	-0.005	0.016
Leverage	(0.029)	(0.026)	(0.022)	(0.024)	(0.027)	(0.025)	(0.022)
Long-term financial debts	-0.004	-0.007	-0.001	-0.004	-0.001	-0.003	0.015
over (equity+debts)	(0.003)	(0.006)	(0.005)	(0.004)	(0.004)	(0.005)	(0.021)
Average east of debt	-0.015	-0.044	-0.015	-0.026	-0.027	-0.094	-0.003
Average cost of debt	(0.037)	(0.052)	(0.050)	(0.034)	(0.055)	(0.069)	(0.003)
ROE	-0.47	-0.49	-0.30	-0.14	-0.05	-0.18	-0.14
ROE	(0.36)	(0.37)	(0.26)	(0.26)	(0.17)	(0.18)	(0.26)
DOA	-0.03	-0.01	-0.02	-0.02	-0.02	-0.02	-0.02
ROA	(0.03)	(0.02)	(0.02)	(0.01)	(0.02)	(0.02)	(0.01)
ROI	-0.16	-0.07	-0.05	-0.07	-0.03	-0.04	-0.07
KUI	(0.12)	(0.05)	(0.07)	(0.06)	(0.09)	(0.11)	(0.06)
# observations	278	276	170	1128	230	174	1128

The size variables and the raw level of investments are measured in thousand euros.

Clustered standard errors by firm included between brackets.

\*,\*\*,\*\*\* for significance levels respectively of 90%, 95% and 99%.

Table 12: Mean differences in firm characteristics in the pre-treatment period (2000) after matching for firms with low credit rating, small size, big size or high subsidy to investments ratio.

	Low crea	dit rating	Small firms	Big	firms	High subsidy to investments ratio		
Measure	Α	В	A B	Α	В	Α	В	
Intangible fixed assets	-17.114	29.280	9.495 -3.164	-21.291	-25.410	-13.767	1.478	
Intangible lixed assets	(0.433)	(0.318)	(0.253) (0.852	(0.141)	(0.117)	(0.141)	(0.963)	
Tangible fixed assets	-14.743	54.640	-16.762 -32.43	6 73.245	-53.988	-42.293	-116.754	
Tangible fixed assets	(0.327)	(0.467)	(0.847) (0.738		(0.449)	(0.700)	(0.614)	
Sales	60.457	45.400	129.564 -9.782		-135.976	83.284	-29.739	
Cales	(0.954)	(0.212)	(0.222) (0.944	, , ,	(0.136)	(0.696)	(0.355)	
Total assets	-30.229	19.000	50.545 -60.09		-33.639	60.155	-14.261	
	(0.969)	(0.386)	(0.202) (0.311	, , ,	(0.135)	(0.469)	(0.402)	
Intangible investments	2.371	-1.400	-6.455 5.782		2.602	8.267	42.652	
-	(0.786)	(0.912)	(0.202) (0.193	, , ,	(0.895)	(0.106)	(0.126)	
Intangible investments over		0.027	-0.004 0.027		0.001	0.004	0.012	
sales	(0.698)	(0.130)	(0.105) (0.141	, , ,	(0.736)	(0.292)	(0.307)	
Intangible investments over		0.015	-0.007 0.008		0.003	0.000	0.015	
total assets	(0.448)	(0.236)	(0.126) (0.452	· · · ·	(0.577)	(0.957)	(0.209)	
Intangible investments over		-0.978	-0.574 -0.096		0.718	-0.534	2.893	
intangible fixed assets	(0.310)	(0.209)	(0.229) (0.605	, , ,	(0.628)	(0.307)	(0.169)	
Tangible investments	-55.400	15.720	-28.485 -42.58		-3.627	1.103	-1.551	
•	(0.189)	(0.430)	(0.197) (0.124	, , ,	(0.942)	(0.970)	(0.964)	
Tangible investments over	-0.055	-0.070	-0.032 -0.042		0.005	0.002	-0.038	
sales	(0.212)	(0.181)	(0.158) (0.116		(0.612)	(0.918)	(0.110)	
Tangible investments over	-0.046	-0.029	-0.036 -0.06		0.010	-0.003	-0.019	
total assets	(0.166)	(0.144)	(0.292) (0.122		(0.453)	(0.807)	(0.303)	
Tangible investments over	-0.181	0.036	-0.208 -0.183		0.010	-0.092	-0.019	
tangible fixed assets	(0.348) 0.003	(0.858)	(0.188) (0.541 0.036 -0.025	, , ,	(0.918) -0.013	(0.394) 0.026	(0.897)	
Value added per employee		-0.042			-0.013 (0.493)		-0.031	
Net working capital over	(0.922) -0.085	(0.318) -0.087	(0.103) (0.456 0.011 -0.076	, , ,	0.045	(0.126) -0.061	(0.159) 0.006	
sales	(0.186)	(0.155)	(0.846) (0.186		(0.170)	(0.105)	(0.876)	
Net working capital over	0.008	-0.079	0.006 -0.022	, , ,	0.042	-0.035	0.022	
total assets	(0.874)	(0.112)	(0.867) (0.668		(0.254)	(0.249)	(0.619)	
	0.005	-0.016	-0.019 0.006	, , ,	-0.001	-0.009	-0.008	
Cash-flow over sales	(0.693)	(0.266)	(0.101) (0.708		(0.873)	(0.134)	(0.408)	
		-0.020	-0.003 0.006		0.005	-0.002	-0.002	
Cash-flow over total assets	(0.622)	(0.162)	(0.200) (0.622		(0.546)	(0.205)	(0.826)	
	-0.039	0.027	0.027 0.012		0.016	0.030	0.011	
Leverage	(0.139)	(0.297)	(0.228) (0.700		(0.497)	(0.129)	(0.668)	
Long-term financial debts	-0.070	-0.053	0.007 -0.05		0.029	-0.001	0.004	
over (equity+debts)	(0.224)	(0.166)	(0.735) (0.140		(0.143)	(0.967)	(0.853)	
	-0.007	0.001	0.003 -0.004	, , ,	0.001	-0.004	0.000	
Average cost of debt	(0.202)	(0.886)	(0.240) (0.318		(0.864)	(0.372)	(0.917)	
	0.895	-0.273	-0.227 0.173	, , ,	-0.040	0.299	0.011	
ROE	(0.220)	(0.432)	(0.138) (0.122		(0.697)	(0.230)	(0.860)	
	-0.006	-0.022	-0.027 -0.014		0.023	-0.030	0.008	
ROA	(0.591)	(0.124)	(0.302) (0.277		(0.175)	(0.101)	(0.630)	
	0.489	-0.232	-0.030 -0.314		0.008	0.174	-0.092	
ROI	(0.129)	(0.199)	(0.336) (0.120		(0.754)	(0.119)	(0.291)	
# observations	70	50	202 110	222	`168 <i>´</i>	232	138	

The size variables and the raw level of investments are measured in thousand euros. Significance level reported between brackets.

The denominators of the investments ratios are measured in the pre-pre-treatment year (1999).

Table 13a: Difference-in-differences results during the project and after for Measure A (firms with low credit rating, small size, big size or high subsidy to investments ratio): net investments

	Years during the project			Average Years after the project					Overall
	1	2	3	project	1	2	3	4	average
Low credit rating									
Intangible investments over	-0.97	-0.16	1.55	-0.37	1.26	-1.58	-1.89	-1.99	-0.61
intangible fixed assets	(0.84)	(0.75)	(3.30)	(1.48)	(2.27)	(2.58)	(2.23)	(2.22)	(1.95)
Tangible investments over	0.08	0.38	0.43	0.24	-0.75	-0.60	0.59	-0.79	-0.11
tangible fixed assets	(0.42)	(1.09)	(0.28)	(0.50)	(0.49)	(0.70)	(0.55)	(1.05)	(0.45)
# observations	70	60	16	146	56	52	42	26	322
	_		Sma	I size	_				_
Intangible investments over		2.14	0.44	1.32	1.12	-0.97	-1.13	-2.63	0.22
intangible fixed assets	(1.44)	(2.11)	(4.08)	(1.34)	(2.57)	(1.71)	(3.04)	(4.32)	(1.57)
Tangible investments over	0.46**	0.34**	0.64**	0.43**	-0.23	-0.23	-0.10	-0.39	0.06
tangible fixed assets	(0.23)	(0.17)	(0.31)	(0.18)	(0.39)	(0.47)	(0.44)	(0.57)	(0.45)
# observations	202	142	40	384	152	140	112	84	872
	-			size					-
Intangible investments over		-0.04	5.01	0.91	2.34	-3.07	-1.62	0.12	0.17
intangible fixed assets	(0.97)	(1.41)	(3.85)	(1.26)	(3.29)	(3.63)	(2.48)	(1.82)	(1.61)
Tangible investments over	0.49***	0.01	0.03	0.24*	-0.25	-0.30	-0.23	-0.26	-0.02
tangible fixed assets	(0.15)	(0.21)	(0.17)	(0.13)	(0.17)	(0.25)	(0.23)	(0.18)	(0.11)
# observations	222	182	74	478	148	140	118	96	980
				investments					_
Intangible investments over		2.27	2.42	1.79	-3.79	-3.34	-1.97	-2.55	-0.79
intangible fixed assets	(1.41)	(1.69)	(4.01)	(1.30)	(3.41)	(2.11)	(1.62)	(3.27)	(1.29)
Tangible investments over	0.55*	0.23	0.15	0.37**	-0.20	-0.36	-0.33	-0.33	0.01
tangible fixed assets	(0.32)	(0.16)	(0.20)	(0.19)	(0.21)	(0.32)	(0.51)	(0.45)	(0.23)
# observations	232	186	70	488	156	148	122	94	1008

Clustered standard errors by firm included between brackets.

The denominators of the investments ratios are measured in the pre-pre-treatment year (1999).

\*,\*\*,\*\*\* for significance levels of the coefficients respectively of 90%, 95% and 99%.

Table 13b: Difference-in-differences results during the project and after for Measure B (firms with low credit rating, small size, big size or high subsidy to investments ratio): net investments

	Yea	rs during project	the	Average Years after during the the project			Overall			
	1	2	3	project	1	2	average			
Low credit rating										
Intangible investments over	3.01*	4.08	3.62**	3.56**	-0.96	1.45	2.31			
intangible fixed assets	(1.63)	(2.52)	(1.71)	(1.61)	(1.84)	(1.14)	(1.47)			
Tangible investments over	0.16	0.51	2.25	0.76	0.28	0.88	0.68			
tangible fixed assets	(0.72)	(0.79)	(1.88)	(0.55)	(0.46)	(0.92)	(0.49)			
# observations	<u>່</u> 50 ໌	50	28	128	40	34	202			
Small size										
Intangible investments over	6.12*	2.32	5.23*	4.47*	-0.35	0.97	3.02			
intangible fixed assets	(3.26)	(3.60)	(3.15)	(2.31)	(3.84)	(2.92)	(2.39)			
Tangible investments over	0.01	1.55	0.49	0.70	0.50	-1.07	0.41			
tangible fixed assets	(0.56)	(1.89)	(0.65)	(0.82)	(0.43)	(3.24)	(0.96)			
# observations	110	110	74	294	88	64	446			
		Big	j size							
Intangible investments over	1.97	2.95	0.45	2.01	3.28	5.52	2.84			
intangible fixed assets	(2.84)	(3.13)	(2.43)	(2.47)	(3.07)	(6.88)	(2.90)			
Tangible investments over	0.13	-0.14	0.35	0.07	-0.11	-0.22	-0.01			
tangible fixed assets	(0.25)	(0.28)	(0.60)	(0.22)	(0.26)	(0.54)	(0.19)			
# observations	168	166	96	430	142	110	682			
High subsidy to investments ratio										
Intangible investments over	1.61	4.79	1.94	2.88	1.99	3.02	2.73			
intangible fixed assets	(9.24)	(4.52)	(4.98)	(4.61)	(2.86)	(3.72)	(3.13)			
Tangible investments over	0.05	-0.57	0.67	-0.02	0.11	0.35	0.06			
tangible fixed assets	(0.36)	(0.36)	(2.20)	(0.60)	(0.44)	(0.65)	(0.47)			
# observations	138	138	94	370	108	84	562			

Clustered standard errors by firm included between brackets.

The denominators of the investments ratios are measured in the pre-pre-treatment year (1999).

\*,\*\*,\*\*\* for significance levels of the coefficients respectively of 90%, 95% and 99%.

#### References

- Almus, M., and Czarnitzki, D. (2003), "The Effects of Public R&D Subsidies on Firms' Innovation Activities: The Case of Eastern Germany", *Journal of Business and Economic Statistics*, 21(2), 226-236
- Arrow, K. J. (1962), "Economic Welfare and the Allocation of Resources for Invention" in Nelson, R. (ed.), *The Rate and Direction of Inventive Activity*, Princeton University Press
- Blundell, R., and Costa Dias, M. (2000), "Evaluation Methods for Non-Experimental Data", *Fiscal Studies*, 21(4), 427-468
- Bronzini, R., and de Blasio, G. (2006), "Evaluating the Impact of Investment Incentives: the Case of Italy's Law 488/1992", *Journal of Urban Economics*, 60, 327-349
- Bronzini, R., and Iachini, E. (2011), Are Incentives for R&D Effective? Evidence from a Regression Discontinuity Approach, Temi di Discussione n. 791, Banca d'Italia
- Bronzini, R., and Piselli, P. (2012), The Impact of R&D Subsidies on Firm Innovation, mimeo
- Busom, I. (2000), "An Empirical Evaluation of the Effects of R&D Subsidies", *Economics of Innovation and New Technology*, 9(2), 111-148
- Carmignani, A., and D'Ignazio, A. (2011), *Financial Subsidies and Bank Lending: Substitutes or Complements? Micro Level Evidence from Italy*, Temi di Discussione n. 803, Banca d'Italia
- Cariola, M., Calabrese, G., and Rolfo, S. (2000), *Principi Teorici ed Evidenze Empiriche nella Valutazione delle Politiche per l'Innovazione delle Regioni Italiane*, III Convegno Nazionale AIV, 23-25/3/2000, Torino
- Dehejia, R. H., and Wahba, S. (2002), "Score-Matching Methods for Non-Experimental Causal Studies", *The Review of Economics and Statistics*, 84(1), 151-161
- Duguet, E. (2003), Are Subsidies a Substitute or a Complement to Privately Funded R&D? Evidence from France using Propensity Score Methods for Non-experimental Data, Working Paper No. 75/2003, Université de Paris I
- Evangelista, R. (2007), "Rilevanza e Impatto delle Politiche d'Innovazione in Italia. I Risultati delle Indagini CIS", *Economia e politica industriale*, 1, 103-124
- Frolich, M. (2004), "Finite-Sample Properties of Propensity-Score Matching and Weighting Estimators", *The Review of Economics and Statistics*, 86(1), 77-90
- Gabriele, R., Zamarian, M., Zaninotto, E. (2007), Assessing the Economic Impact of Public Industrial Policies: An Empirical Investigation on Subsidies, working paper
- Hall, B. H. (2002), "The Financing of Research and Development", *Oxford Review of Economic Policy*, 18(1), 35-51
- Hatzichronoglou, T. (1997), *Revision of the High-Technology Sector and Product Classification*, OECD Science, Technology and Industry Working Papers 1997/2, OECD
- Hussinger, K. (2008), "R&D and Subsidies at the Firm Level: An Application of Parametric and Semi-Parametric Two-Step Selection Models", *Journal of Applied Econometrics*, 23, 729-747

- IMF (2003), External Debt Statistics: Guide for Compilers and Users, IMF
- Lach, S. (2002), "Do R&D Subsidies Stimulate or Displace Private R&D? Evidence from Israel", *The Journal of Industrial Economics*, 50(4), 369-390
- Lööf, H., and Heshmati, A. (2007), "The Impact of Public Funds on Private R&D Investment: New Evidence from a Firm Level Innovation Study", in Heshmati A., Sohn, Y-B., and Kim, Y-R. (eds.), *Commercialization and Transfer of Technology: Major Country Case Studies*, Nova Science Publishers
- OECD (2007), OECD Science, Technology and Industry Scoreboard, OECD
- Wallsten, S. J. (2000), "The Effect of Government-industry R&D Programs on Private R&D: the Case of Small Business Innovation Research Program", *RAND Journal of Economics*, 31(1), 82-100

#### RECENTLY PUBLISHED "TEMI" (\*)

- N. 878 Do food commodity prices have asymmetric effects on euro-area inflation?, by Mario Porqueddu and Fabrizio Venditti (September 2012).
- N. 879 Industry dynamics and competition from low-wage countries: evidence on Italy, by Stefano Federico (September 2012).
- N. 880 *The micro dynamics of exporting: evidence from French firms*, by Ines Buono and Harald Fadinger (September 2012).
- N. 881 On detecting end-of-sample instabilities, by Fabio Busetti (September 2012).
- N. 882 An empirical comparison of alternative credit default swap pricing models, by Michele Leonardo Bianchi (September 2012).
- N. 883 Learning, incomplete contracts and export dynamics: theory and evidence from French firms, by Romain Aeberhardt, Ines Buono and Harald Fadinger (October 2012).
- N. 884 Collaboration between firms and universities in Italy: the role of a firm's proximity to top-rated departments, by Davide Fantino, Alessandra Mori and Diego Scalise (October 2012).
- N. 885 *Parties, institutions and political budget cycles at the municipal level*, by Marika Cioffi, Giovanna Messina and Pietro Tommasino (October 2012).
- N. 886 Immigration, jobs and employment protection: evidence from Europe before and during the Great Recession, by Francesco D'Amuri and Giovanni Peri (October 2012).
- N. 887 A structural model for the housing and credit markets in Italy, by Andrea Nobili and Francesco Zollino (October 2012).
- N. 888 Monetary policy in a model with misspecified, heterogeneous and ever-changing expectations, by Alberto Locarno (October 2012).
- N. 889 Does corporate taxation affect cross-country firm leverage?, by Antonio De Socio and Valentina Nigro (November 2012).
- N. 890 *Start-up banks' default and the role of capital*, by Massimo Libertucci and Francesco Piersante (November 2012).
- N. 891 The predictive power of Google searches in forecasting unemployment, by Francesco D'Amuri and Juri Marcucci (November 2012).
- N. 892 A dynamic default dependence model, by Sara Cecchetti and Giovanna Nappo (November 2012).
- N. 893 *Externalities in interbank network: results from a dynamic simulation model*, by Michele Manna and Alessandro Schiavone (November 2012).
- N. 894 Incentives through the cycle: microfounded macroprudential regulation, by Giovanni di Iasio and Mario Quagliariello (January 2013).
- N. 895 What is a prime bank? A Euribor OIS spread perspective, by Marco Taboga (January 2013).
- N. 896 Short-term GDP forecasting with a mixed frequency dynamic factor model with stochastic volatility, by Massimiliano Marcellino, Mario Porqueddu and Fabrizio Venditti (January 2013).
- N. 897 Collective action clauses: how do they weigh on sovereigns?, by Alfredo Bardozzetti and Davide Dottori (January 2013).
- N. 898 Firm size and judicial efficiency: evidence from the neighbour's Court, by Silvia Giacomelli and Carlo Menon (January 2013).
- N. 899 A spatial competitive analysis: the carbon leakage effect on the cement industry under the European Emissions Trading Scheme, by Elisabetta Allevi, Giorgia Oggioni, Rossana Riccardi and Marco Rocco (January 2013).

<sup>(\*)</sup> Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

- F. PANETTA, F. SCHIVARDI and M. SHUM, *Do mergers improve information? Evidence from the loan market*, Journal of Money, Credit, and Banking, v. 41, 4, pp. 673-709, **TD No. 521 (October 2004).**
- M. BUGAMELLI and F. PATERNÒ, *Do workers' remittances reduce the probability of current account reversals?*, World Development, v. 37, 12, pp. 1821-1838, **TD No. 573 (January 2006).**
- P. PAGANO and M. PISANI, *Risk-adjusted forecasts of oil prices*, The B.E. Journal of Macroeconomics, v. 9, 1, Article 24, **TD No. 585 (March 2006).**
- M. PERICOLI and M. SBRACIA, The CAPM and the risk appetite index: theoretical differences, empirical similarities, and implementation problems, International Finance, v. 12, 2, pp. 123-150, TD No. 586 (March 2006).
- R. BRONZINI and P. PISELLI, *Determinants of long-run regional productivity with geographical spillovers: the role of R&D, human capital and public infrastructure,* Regional Science and Urban Economics, v. 39, 2, pp.187-199, **TD No. 597 (September 2006).**
- U. ALBERTAZZI and L. GAMBACORTA, *Bank profitability and the business cycle*, Journal of Financial Stability, v. 5, 4, pp. 393-409, **TD No. 601 (September 2006).**
- F. BALASSONE, D. FRANCO and S. ZOTTERI, *The reliability of EMU fiscal indicators: risks and safeguards*, in M. Larch and J. Nogueira Martins (eds.), Fiscal Policy Making in the European Union: an Assessment of Current Practice and Challenges, London, Routledge, **TD No. 633 (June 2007).**
- A. CIARLONE, P. PISELLI and G. TREBESCHI, *Emerging Markets' Spreads and Global Financial Conditions*, Journal of International Financial Markets, Institutions & Money, v. 19, 2, pp. 222-239, **TD No. 637 (June 2007)**.
- S. MAGRI, *The financing of small innovative firms: the Italian case*, Economics of Innovation and New Technology, v. 18, 2, pp. 181-204, **TD No. 640 (September 2007).**
- V. DI GIACINTO and G. MICUCCI, The producer service sector in Italy: long-term growth and its local determinants, Spatial Economic Analysis, Vol. 4, No. 4, pp. 391-425, TD No. 643 (September 2007).
- F. LORENZO, L. MONTEFORTE and L. SESSA, *The general equilibrium effects of fiscal policy: estimates for the euro area*, Journal of Public Economics, v. 93, 3-4, pp. 559-585, **TD No. 652** (November 2007).
- Y. ALTUNBAS, L. GAMBACORTA and D. MARQUÉS, *Securitisation and the bank lending channel*, European Economic Review, v. 53, 8, pp. 996-1009, **TD No. 653** (November 2007).
- R. GOLINELLI and S. MOMIGLIANO, *The Cyclical Reaction of Fiscal Policies in the Euro Area. A Critical Survey of Empirical Research*, Fiscal Studies, v. 30, 1, pp. 39-72, **TD No. 654 (January 2008).**
- P. DEL GIOVANE, S. FABIANI and R. SABBATINI, What's behind "Inflation Perceptions"? A survey-based analysis of Italian consumers, Giornale degli Economisti e Annali di Economia, v. 68, 1, pp. 25-52, TD No. 655 (January 2008).
- F. MACCHERONI, M. MARINACCI, A. RUSTICHINI and M. TABOGA, *Portfolio selection with monotone mean*variance preferences, Mathematical Finance, v. 19, 3, pp. 487-521, **TD No. 664 (April 2008).**
- M. AFFINITO and M. PIAZZA, What are borders made of? An analysis of barriers to European banking integration, in P. Alessandrini, M. Fratianni and A. Zazzaro (eds.): The Changing Geography of Banking and Finance, Dordrecht Heidelberg London New York, Springer, TD No. 666 (April 2008).
- A. BRANDOLINI, On applying synthetic indices of multidimensional well-being: health and income inequalities in France, Germany, Italy, and the United Kingdom, in R. Gotoh and P. Dumouchel (eds.), Against Injustice. The New Economics of Amartya Sen, Cambridge, Cambridge University Press, TD No. 668 (April 2008).
- G. FERRERO and A. NOBILI, *Futures contract rates as monetary policy forecasts*, International Journal of Central Banking, v. 5, 2, pp. 109-145, **TD No. 681 (June 2008).**
- P. CASADIO, M. LO CONTE and A. NERI, Balancing work and family in Italy: the new mothers' employment decisions around childbearing, in T. Addabbo and G. Solinas (eds.), Non-Standard Employment and Qualità of Work, Physica-Verlag. A Sprinter Company, TD No. 684 (August 2008).
- L. ARCIERO, C. BIANCOTTI, L. D'AURIZIO and C. IMPENNA, *Exploring agent-based methods for the analysis* of payment systems: A crisis model for StarLogo TNG, Journal of Artificial Societies and Social Simulation, v. 12, 1, **TD No. 686 (August 2008).**
- A. CALZA and A. ZAGHINI, Nonlinearities in the dynamics of the euro area demand for M1, Macroeconomic Dynamics, v. 13, 1, pp. 1-19, **TD No. 690 (September 2008).**
- L. FRANCESCO and A. SECCHI, *Technological change and the households' demand for currency*, Journal of Monetary Economics, v. 56, 2, pp. 222-230, **TD No. 697 (December 2008).**

- G. ASCARI and T. ROPELE, *Trend inflation, taylor principle, and indeterminacy*, Journal of Money, Credit and Banking, v. 41, 8, pp. 1557-1584, **TD No. 708** (May 2007).
- S. COLAROSSI and A. ZAGHINI, *Gradualism, transparency and the improved operational framework: a look at overnight volatility transmission*, International Finance, v. 12, 2, pp. 151-170, **TD No. 710 (May 2009).**
- M. BUGAMELLI, F. SCHIVARDI and R. ZIZZA, *The euro and firm restructuring*, in A. Alesina e F. Giavazzi (eds): Europe and the Euro, Chicago, University of Chicago Press, **TD No. 716 (June 2009).**
- B. HALL, F. LOTTI and J. MAIRESSE, *Innovation and productivity in SMEs: empirical evidence for Italy*, Small Business Economics, v. 33, 1, pp. 13-33, **TD No. 718 (June 2009).**

- A. PRATI and M. SBRACIA, Uncertainty and currency crises: evidence from survey data, Journal of Monetary Economics, v, 57, 6, pp. 668-681, **TD No. 446 (July 2002).**
- L. MONTEFORTE and S. SIVIERO, *The Economic Consequences of Euro Area Modelling Shortcuts*, Applied Economics, v. 42, 19-21, pp. 2399-2415, **TD No. 458 (December 2002).**
- S. MAGRI, *Debt maturity choice of nonpublic Italian firms*, Journal of Money, Credit, and Banking, v.42, 2-3, pp. 443-463, **TD No. 574 (January 2006).**
- G. DE BLASIO and G. NUZZO, *Historical traditions of civicness and local economic development*, Journal of Regional Science, v. 50, 4, pp. 833-857, **TD No. 591 (May 2006).**
- E. IOSSA and G. PALUMBO, *Over-optimism and lender liability in the consumer credit market*, Oxford Economic Papers, v. 62, 2, pp. 374-394, **TD No. 598 (September 2006).**
- S. NERI and A. NOBILI, *The transmission of US monetary policy to the euro area,* International Finance, v. 13, 1, pp. 55-78, **TD No. 606 (December 2006).**
- F. ALTISSIMO, R. CRISTADORO, M. FORNI, M. LIPPI and G. VERONESE, *New Eurocoin: Tracking Economic Growth in Real Time*, Review of Economics and Statistics, v. 92, 4, pp. 1024-1034, **TD No. 631 (June 2007).**
- U. ALBERTAZZI and L. GAMBACORTA, *Bank profitability and taxation*, Journal of Banking and Finance, v. 34, 11, pp. 2801-2810, **TD No. 649** (November 2007).
- M. IACOVIELLO and S. NERI, *Housing market spillovers: evidence from an estimated DSGE model,* American Economic Journal: Macroeconomics, v. 2, 2, pp. 125-164, **TD No. 659 (January 2008).**
- F. BALASSONE, F. MAURA and S. ZOTTERI, *Cyclical asymmetry in fiscal variables in the EU*, Empirica, **TD** No. 671, v. 37, 4, pp. 381-402 (June 2008).
- F. D'AMURI, O. GIANMARCO I.P. and P. GIOVANNI, The labor market impact of immigration on the western german labor market in the 1990s, European Economic Review, v. 54, 4, pp. 550-570, TD No. 687 (August 2008).
- A. ACCETTURO, Agglomeration and growth: the effects of commuting costs, Papers in Regional Science, v. 89, 1, pp. 173-190, **TD No. 688 (September 2008).**
- S. NOBILI and G. PALAZZO, *Explaining and forecasting bond risk premiums*, Financial Analysts Journal, v. 66, 4, pp. 67-82, **TD No. 689 (September 2008).**
- A. B. ATKINSON and A. BRANDOLINI, *On analysing the world distribution of income*, World Bank Economic Review, v. 24, 1, pp. 1-37, **TD No. 701 (January 2009).**
- R. CAPPARIELLO and R. ZIZZA, Dropping the Books and Working Off the Books, Labour, v. 24, 2, pp. 139-162, **TD No. 702 (January 2009).**
- C. NICOLETTI and C. RONDINELLI, *The (mis)specification of discrete duration models with unobserved heterogeneity: a Monte Carlo study*, Journal of Econometrics, v. 159, 1, pp. 1-13, **TD No. 705** (March 2009).
- L. FORNI, A. GERALI and M. PISANI, *Macroeconomic effects of greater competition in the service sector: the case of Italy*, Macroeconomic Dynamics, v. 14, 5, pp. 677-708, **TD No. 706** (March 2009).
- V. DI GIACINTO, G. MICUCCI and P. MONTANARO, Dynamic macroeconomic effects of public capital: evidence from regional Italian data, Giornale degli economisti e annali di economia, v. 69, 1, pp. 29-66, TD No. 733 (November 2009).
- F. COLUMBA, L. GAMBACORTA and P. E. MISTRULLI, *Mutual Guarantee institutions and small business finance*, Journal of Financial Stability, v. 6, 1, pp. 45-54, **TD No. 735** (November 2009).
- A. GERALI, S. NERI, L. SESSA and F. M. SIGNORETTI, *Credit and banking in a DSGE model of the Euro Area,* Journal of Money, Credit and Banking, v. 42, 6, pp. 107-141, **TD No. 740 (January 2010).**
- M. AFFINITO and E. TAGLIAFERRI, Why do (or did?) banks securitize their loans? Evidence from Italy, Journal

of Financial Stability, v. 6, 4, pp. 189-202, TD No. 741 (January 2010).

- S. FEDERICO, Outsourcing versus integration at home or abroad and firm heterogeneity, Empirica, v. 37, 1, pp. 47-63, **TD No. 742** (February 2010).
- V. DI GIACINTO, *On vector autoregressive modeling in space and time*, Journal of Geographical Systems, v. 12, 2, pp. 125-154, **TD No. 746 (February 2010).**
- L. FORNI, A. GERALI and M. PISANI, *The macroeconomics of fiscal consolidations in euro area countries,* Journal of Economic Dynamics and Control, v. 34, 9, pp. 1791-1812, **TD No. 747** (March 2010).
- S. MOCETTI and C. PORELLO, *How does immigration affect native internal mobility? new evidence from Italy*, Regional Science and Urban Economics, v. 40, 6, pp. 427-439, **TD No. 748 (March 2010)**.
- A. DI CESARE and G. GUAZZAROTTI, An analysis of the determinants of credit default swap spread changes before and during the subprime financial turmoil, Journal of Current Issues in Finance, Business and Economics, v. 3, 4, pp., **TD No. 749** (March 2010).
- P. CIPOLLONE, P. MONTANARO and P. SESTITO, Value-added measures in Italian high schools: problems and findings, Giornale degli economisti e annali di economia, v. 69, 2, pp. 81-114, TD No. 754 (March 2010).
- A. BRANDOLINI, S. MAGRI and T. M SMEEDING, *Asset-based measurement of poverty*, Journal of Policy Analysis and Management, v. 29, 2, pp. 267-284, **TD No. 755** (March 2010).
- G. CAPPELLETTI, A Note on rationalizability and restrictions on beliefs, The B.E. Journal of Theoretical Economics, v. 10, 1, pp. 1-11, **TD No. 757** (April 2010).
- S. DI ADDARIO and D. VURI, Entrepreneurship and market size. the case of young college graduates in *Italy*, Labour Economics, v. 17, 5, pp. 848-858, **TD No. 775 (September 2010).**
- A. CALZA and A. ZAGHINI, *Sectoral money demand and the great disinflation in the US*, Journal of Money, Credit, and Banking, v. 42, 8, pp. 1663-1678, **TD No. 785 (January 2011).**

- S. DI ADDARIO, *Job search in thick markets*, Journal of Urban Economics, v. 69, 3, pp. 303-318, **TD No.** 605 (December 2006).
- F. SCHIVARDI and E. VIVIANO, *Entry barriers in retail trade*, Economic Journal, v. 121, 551, pp. 145-170, **TD** No. 616 (February 2007).
- G. FERRERO, A. NOBILI and P. PASSIGLIA, Assessing excess liquidity in the Euro Area: the role of sectoral distribution of money, Applied Economics, v. 43, 23, pp. 3213-3230, **TD No. 627** (April 2007).
- P. E. MISTRULLI, Assessing financial contagion in the interbank market: maximum entropy versus observed interbank lending patterns, Journal of Banking & Finance, v. 35, 5, pp. 1114-1127, TD No. 641 (September 2007).
- E. CIAPANNA, Directed matching with endogenous markov probability: clients or competitors?, The RAND Journal of Economics, v. 42, 1, pp. 92-120, **TD No. 665 (April 2008).**
- M. BUGAMELLI and F. PATERNÒ, *Output growth volatility and remittances*, Economica, v. 78, 311, pp. 480-500, **TD No. 673 (June 2008).**
- V. DI GIACINTO e M. PAGNINI, Local and global agglomeration patterns: two econometrics-based indicators, Regional Science and Urban Economics, v. 41, 3, pp. 266-280, **TD No. 674 (June 2008)**.
- G. BARONE and F. CINGANO, Service regulation and growth: evidence from OECD countries, Economic Journal, v. 121, 555, pp. 931-957, TD No. 675 (June 2008).
- R. GIORDANO and P. TOMMASINO, What determines debt intolerance? The role of political and monetary *institutions*, European Journal of Political Economy, v. 27, 3, pp. 471-484, **TD No. 700 (January 2009).**
- P. ANGELINI, A. NOBILI e C. PICILLO, *The interbank market after August 2007: What has changed, and why?*, Journal of Money, Credit and Banking, v. 43, 5, pp. 923-958, **TD No. 731 (October 2009).**
- L. FORNI, A. GERALI and M. PISANI, *The Macroeconomics of Fiscal Consolidation in a Monetary Union: the Case of Italy*, in Luigi Paganetto (ed.), Recovery after the crisis. Perspectives and policies, VDM Verlag Dr. Muller, **TD No. 747 (March 2010).**
- A. DI CESARE and G. GUAZZAROTTI, An analysis of the determinants of credit default swap changes before and during the subprime financial turmoil, in Barbara L. Campos and Janet P. Wilkins (eds.), The Financial Crisis: Issues in Business, Finance and Global Economics, New York, Nova Science Publishers, Inc., **TD No. 749 (March 2010).**
- A. LEVY and A. ZAGHINI, *The pricing of government guaranteed bank bonds*, Banks and Bank Systems, v. 6, 3, pp. 16-24, **TD No. 753 (March 2010).**

- G. GRANDE and I. VISCO, A public guarantee of a minimum return to defined contribution pension scheme members, The Journal of Risk, v. 13, 3, pp. 3-43, **TD No. 762 (June 2010).**
- P. DEL GIOVANE, G. ERAMO and A. NOBILI, *Disentangling demand and supply in credit developments: a survey-based analysis for Italy*, Journal of Banking and Finance, v. 35, 10, pp. 2719-2732, **TD No.** 764 (June 2010).
- G. BARONE and S. MOCETTI, With a little help from abroad: the effect of low-skilled immigration on the female labour supply, Labour Economics, v. 18, 5, pp. 664-675, **TD No. 766 (July 2010).**
- A. FELETTIGH and S. FEDERICO, *Measuring the price elasticity of import demand in the destination markets of italian exports*, Economia e Politica Industriale, v. 38, 1, pp. 127-162, **TD No. 776 (October 2010).**
- S. MAGRI and R. PICO, *The rise of risk-based pricing of mortgage interest rates in Italy*, Journal of Banking and Finance, v. 35, 5, pp. 1277-1290, **TD No. 778 (October 2010).**
- M. TABOGA, Under/over-valuation of the stock market and cyclically adjusted earnings, International Finance, v. 14, 1, pp. 135-164, **TD No. 780 (December 2010).**
- S. NERI, *Housing, consumption and monetary policy: how different are the U.S. and the Euro area?*, Journal of Banking and Finance, v.35, 11, pp. 3019-3041, **TD No. 807** (April 2011).
- V. CUCINIELLO, *The welfare effect of foreign monetary conservatism with non-atomistic wage setters*, Journal of Money, Credit and Banking, v. 43, 8, pp. 1719-1734, **TD No. 810 (June 2011).**
- A. CALZA and A. ZAGHINI, welfare costs of inflation and the circulation of US currency abroad, The B.E. Journal of Macroeconomics, v. 11, 1, Art. 12, **TD No. 812 (June 2011).**
- I. FAIELLA, *La spesa energetica delle famiglie italiane*, Energia, v. 32, 4, pp. 40-46, **TD No. 822 (September 2011).**
- R. DE BONIS and A. SILVESTRINI, The effects of financial and real wealth on consumption: new evidence from OECD countries, Applied Financial Economics, v. 21, 5, pp. 409–425, TD No. 837 (November 2011).

- F. CINGANO and A. ROSOLIA, *People I know: job search and social networks*, Journal of Labor Economics, v. 30, 2, pp. 291-332, **TD No. 600 (September 2006).**
- G. GOBBI and R. ZIZZA, Does the underground economy hold back financial deepening? Evidence from the italian credit market, Economia Marche, Review of Regional Studies, v. 31, 1, pp. 1-29, TD No. 646 (November 2006).
- S. MOCETTI, *Educational choices and the selection process before and after compulsory school*, Education Economics, v. 20, 2, pp. 189-209, **TD No. 691 (September 2008).**
- F. LIPPI and A. NOBILI, *Oil and the macroeconomy: a quantitative structural analysis,* Journal of European Economic Association, v. 10, 5, pp. 1059-1083, **TD No. 704** (March 2009).
- S. FEDERICO, *Headquarter intensity and the choice between outsourcing versus integration at home or abroad*, Industrial and Corporate Chang, v. 21, 6, pp. 1337-1358, **TD No. 742 (February 2010).**
- S. GOMES, P. JACQUINOT and M. PISANI, The EAGLE. A model for policy analysis of macroeconomic interdependence in the euro area, Economic Modelling, v. 29, 5, pp. 1686-1714, TD No. 770 (July 2010).
- A. ACCETTURO and G. DE BLASIO, Policies for local development: an evaluation of Italy's "Patti Territoriali", Regional Science and Urban Economics, v. 42, 1-2, pp. 15-26, TD No. 789 (January 2006).
- F. BUSETTI and S. DI SANZO, *Bootstrap LR tests of stationarity, common trends and cointegration,* Journal of Statistical Computation and Simulation, v. 82, 9, pp. 1343-1355, **TD No. 799 (March 2006).**
- S. NERI and T. ROPELE, *Imperfect information, real-time data and monetary policy in the Euro area,* The Economic Journal, v. 122, 561, pp. 651-674, **TD No. 802 (March 2011).**
- G. CAPPELLETTI, G. GUAZZAROTTI and P. TOMMASINO, *What determines annuity demand at retirement?*, The Geneva Papers on Risk and Insurance – Issues and Practice, pp. 1-26, **TD No. 808 (April 2011).**
- A. ANZUINI and F. FORNARI, *Macroeconomic determinants of carry trade activity*, Review of International Economics, v. 20, 3, pp. 468-488, **TD No. 817 (September 2011).**
- M. AFFINITO, *Do interbank customer relationships exist? And how did they function in the crisis? Learning from Italy*, Journal of Banking and Finance, v. 36, 12, pp. 3163-3184, **TD No. 826 (October 2011).**
- R. CRISTADORO and D. MARCONI, *Household savings in China*, Journal of Chinese Economic and Business Studies, v. 10, 3, pp. 275-299, **TD No. 838 (November 2011).**

- V. DI GIACINTO, G. MICUCCI and P. MONTANARO, Network effects of public transposrt infrastructure: evidence on Italian regions, Papers in Regional Science, v. 91, 3, pp. 515-541, TD No. 869 (July 2012).
- A. FILIPPIN and M. PACCAGNELLA, *Family background, self-confidence and economic outcomes,* Economics of Education Review, v. 31, 5, pp. 824-834, **TD No. 875 (July 2012).**

#### 2013

F. BUSETTI and J. MARCUCCI, *Comparing forecast accuracy: a Monte Carlo investigation*, International Journal of Forecasting, v. 29, 1, pp. 13-27, **TD No. 723 (September 2009).** 

#### FORTHCOMING

- M. BUGAMELLI and A. ROSOLIA, *Produttività e concorrenza estera*, Rivista di politica economica, **TD No.** 578 (February 2006).
- P. SESTITO and E. VIVIANO, *Reservation wages: explaining some puzzling regional patterns*, Labour, **TD No. 696 (December 2008).**
- P. PINOTTI, M. BIANCHI and P. BUONANNO, *Do immigrants cause crime?*, Journal of the European Economic Association, **TD No. 698 (December 2008).**
- F. CINGANO and P. PINOTTI, *Politicians at work. The private returns and social costs of political connections*, Journal of the European Economic Association, **TD No. 709 (May 2009).**
- Y. ALTUNBAS, L. GAMBACORTA and D. MARQUÉS-IBÁÑEZ, *Bank risk and monetary policy*, Journal of Financial Stability, **TD No. 712 (May 2009).**
- G. BARONE and S. MOCETTI, *Tax morale and public spending inefficiency*, International Tax and Public Finance, **TD No. 732 (November 2009).**
- I. BUONO and G. LALANNE, *The effect of the Uruguay Round on the intensive and extensive margins of trade*, Journal of International Economics, **TD No. 743 (February 2010).**
- G. BARONE, R. FELICI and M. PAGNINI, *Switching costs in local credit markets,* International Journal of Industrial Organization, **TD No. 760 (June 2010).**
- E. COCOZZA and P. PISELLI, Testing for east-west contagion in the European banking sector during the financial crisis, in R. Matoušek; D. Stavárek (eds.), Financial Integration in the European Union, Taylor & Francis, TD No. 790 (February 2011).
- E. GAIOTTI, Credit availablility and investment: lessons from the "Great Recession", European Economic Review, **TD No. 793 (February 2011).**
- A. DE SOCIO, Squeezing liquidity in a "lemons market" or asking liquidity "on tap", Journal of Banking and Finance, **TD No. 819 (September 2011).**
- O. BLANCHARD and M. RIGGI, Why are the 2000s so different from the 1970s? A structural interpretation of changes in the macroeconomic effects of oil prices, Journal of the European Economic Association, **TD No. 835 (November 2011).**
- S. FEDERICO, *Industry dynamics and competition from low-wage countries: evidence on Italy*, Oxford Bulletin of Economics and Statistics, **TD No. 879 (September 2012).**
- F. D'AMURI and G. PERI, Immigration, jobs and employment protection: evidence from Europe before and during the Great Recession, Journal of the European Economic Association, TD No. 886 (October 2012).